CONSTITUTION OF	
PACIFIC EDGE LIMITED	Deleted: BIOTECHNOLOGY

Deleted: Certified as the Constitution of the Compa\_...[1]

1904396 4422040

PART A:	INTRODUCTION	1
1.	DEFINITIONS AND INTERPRETATION	1
	DEFINED TERMS	1
1.2	INTERPRETATION	3
1.3 <b>Use</b>	OF ELECTRONIC MEANS	4
2.	APPLICATION OF THE LISTING RULES	4
2.1	COMPLIANCE WITH LISTING RULES	4
2.2	EFFECT OF RULINGS BY THE NZX	4
2.3	EFFECT OF FAILURE TO COMPLY WITH LISTING RULES	5
PART B:	SHARES AND SHAREHOLDERS	5
3	RIGHTS ATTACHING TO SHARES	5
	CLASSES OF SHARES	5
0.1	CLASSES OF CHARES	0
4.	ISSUE OF SHARES AND OTHER SECURITIES	5
	BOARD HAS POWER TO ISSUE SHARES AND OTHER SECURITIES	5
	PRE-EMPTIVE RIGHTS ON ISSUE	5
4.3	CONSOLIDATION AND SUBDIVISION OF SHARES	7
5.	MODIFICATION OF SHAREHOLDER RIGHTS	8
5.1	PROCEDURE FOR MODIFICATION OF SHAREHOLDER RIGHTS	8
5.2	EXCEPTIONS	8
5.3	ISSUE OF EQUAL RANKING EQUITY SECURITIES	8
5.4	CANCELLATION OF UNPAID AMOUNTS	8
5.5	ACTIONS NOT INVALID	8
6.	CALLS ON SHARES	9
6.1	BOARD MAY MAKE CALLS	9
6.2	TIME OF CALL	9
6.3	FIXED INSTALMENT DEEMED CALLS	9
6.4		9
6.5	DIFFERENT AMOUNTS	9
	LIABILITY	9
	INTEREST AND EXPENSES	9
	PROCEEDINGS FOR RECOVERY OF CALL	10
6.9	PAYMENT IN ADVANCE OF CALLS	10
7.	LIEN ON SHARES	10
7.1	LIEN ON SHARES	10
7.2	WAIVER OF LIEN	10
0		40
8.	FORFEITURE OF SHARES	10
<u>8.1</u> 8.2	NOTICE REQUIRING PAYMENT OF CALL	10
	CONTENTS OF NOTICE	11
8.3	FORFEITURE FOR NON-PAYMENT	11
<u>8.4</u> 8.5	EVIDENCE OF FORFEITURE	11
0.0	CANCELLATION OF FORFEITURE	11

	E		
8.6	EFFECT OF FORFEITURE	11	
9.	SALE OF SHARES SUBJECT TO LIEN OR FORFEITURE	11	
<u>9.1</u>	Power of Sale	11	
9.2	APPLICATION OF PROCEEDS OF SALE	12	
9.3	SALE PROCEDURE	12	
10.	DISTRIBUTIONS TO SHAREHOLDERS	12	
<u>10.1</u>	SATISFACTION OF SOLVENCY TEST	12	
10.2	RESTRICTION ON CERTAIN DISTRIBUTIONS	12	
<u>10.3</u>	METHOD OF PAYMENT	13	
10.4	SHARES IN LIEU OF DIVIDENDS	13	
10.5	Bonus Issues	13	
10.6	SHAREHOLDER DISCOUNTS	14	
10.7	DEDUCTION	14	
10.8	NO INTEREST	14	
10.9	UNCLAIMED DISTRIBUTIONS	14	
11.	ACQUISITION AND REDEMPTION OF COMPANY'S OWN SHARES	15	
11.1	COMPANY MAY PURCHASE, ACQUIRE OR REDEEM ITS OWN SHARES	15	
12.	FINANCIAL ASSISTANCE BY THE COMPANY TO PURCHASE ITS SHA	RES 15	
12.1	FINANCIAL ASSISTANCE	15	
12.2	PERMITTED FINANCIAL ASSISTANCE	15	
12.3	FINANCIAL ASSISTANCE WITH APPROVAL OF EQUITY SECURITY HOLDERS	16	
40		40	
<u>13.</u>	TRANSFER OF SHARES	16	
<u>13.1</u>		16	
13.2	FORM OF TRANSFER	16	
<u>13.3</u> 13.4	TRANSFERS EXECUTED OVERSEAS OTHER FORMS OF TRANSFER	<u>16</u> 16	
13.4	DELIVERY TO COMPANY	16	
13.6	POWER TO REFUSE OR DELAY REGISTRATION	10	
13.7	COMPULSORY DISPOSAL WHEN HOLDING LESS THAN MINIMUM HOLDING	17	
13.8	WHEN TRANSFER EFFECTIVE	18	
13.9	MULTIPLE REGISTERS	18	
13.10	TRANSFER OF RIGHTS	18	
15.10		10	
14.	TRANSMISSIONS AND EQUITABLE INTERESTS	18	
14.1	TRANSMISSION ON DEATH OF SHAREHOLDER	18	
14.2	RIGHTS OF PERSONAL REPRESENTATIVES	18	
14.3	JOINT PERSONAL REPRESENTATIVES	18,	Deleted: 19
14.4	No Notice of Trusts	19	
<u>14.5</u>	NO RECOGNITION OF EQUITABLE INTERESTS	<u>19</u>	
15.	SEPARATE PARCELS OF SHARES	19	
15.1	SEPARATE PARCELS	19	
16.	SHAREHOLDER REPORTS, INFORMATION AND NOTICES	19	
16.1	COMPANY REPORTS AND FINANCIAL STATEMENTS	19	
16.2	INSPECTION BY SHAREHOLDERS	19	
16.3	METHOD OF SERVICE OF NOTICES, REPORTS AND OTHER DOCUMENTS	19	
16.4	SERVICE OF NOTICES OVERSEAS	19	Deleted: 20

<u>16.5</u>	ACCIDENTAL OMISSIONS	20	
<u>16.6</u>	JOINT SHAREHOLDERS	20	
17.	MEETINGS OF SHAREHOLDERS	20	
17.1	ANNUAL MEETINGS	20	
17.2	TIME AND PLACE OF ANNUAL MEETING	20	
17.3	SPECIAL MEETINGS	20	
17.4	CALLING OF SPECIAL MEETINGS	20	
17.5	EQUITY SECURITY HOLDERS ENTITLED TO ATTEND	20	Deleted: 21
17.6	METHODS OF HOLDING MEETINGS	20	
17.0		20,	Deleted: 21
40		~	
	PROCEDURE AT SHAREHOLDER MEETINGS	21	
	THE PROVISIONS OF THE FIRST SCHEDULE TO THIS CONSTITUTION GOVERN	~	
	INGS AT MEETINGS OF SHAREHOLDERS.	21	
	THE PROVISIONS OF THE FIRST SCHEDULE TO THIS CONSTITUTION, WITH ALL		
	RY MODIFICATIONS, GOVERN PROCEEDINGS AT MEETINGS OF INTEREST GROUPS AND		
CLASSES.			
PART C:	DIRECTORS	22	
19.	POWERS OF DIRECTORS	22	
<u>19.</u> 19.1	MANAGEMENT OF COMPANY	22	
	Exercise of Powers by Board	22	
<u>19.2</u>		23	
<u>19.3</u>	DELEGATION OF POWERS	23	
<u>19.4</u>		23	
<u>19.5</u>	RATIFICATION BY SHAREHOLDERS	23	
<u>20.</u>	APPOINTMENT AND REMOVAL OF DIRECTORS	23	
20.1	NUMBER OF DIRECTORS AND RESIDENCE	23	
20.2		23	
20.3	EXISTING DIRECTORS TO CONTINUE	23	
20.4	NOMINATION OF DIRECTORS	23	
20.5	RESTRICTION ON APPOINTMENT OF SEVERAL DIRECTORS BY SINGLE RESOLUTION	24	
20.6	ROTATION OF DIRECTORS	24	
20.7	ROTATION REQUIREMENTS	24	
20.8	VACATION OF OFFICE	24	
20.9	APPOINTMENT BY BOARD	25	
21.	ALTERNATE DIRECTORS	25	
21.1	APPOINTMENT	25	
21.2	RIGHTS OF ALTERNATE DIRECTOR	25	
21.3	REMUNERATION AND EXPENSES	25	
21.4	CESSATION OF APPOINTMENT	25	
21.5	27		
22		27	
22.	DIRECTOR EXPENSES	27	
<u>23.</u>		28	
23.1	INDEMNITY OF DIRECTORS	28	
23.2	OTHER INDEMNITIES	28	
23.3	EXCEPTIONS	29	
23.4		29	
23.5	DEFINITIONS	29	

# 24. PROCEEDINGS OF THE BOARD 29 24.1 THE PROVISIONS OF THE SECOND SCHEDULE TO THIS CONSTITUTION GOVERN THE PROCEEDINGS OF THE BOARD. THE THIRD SCHEDULE TO THE ACT DOES NOT APPLY TO THE COMPANY.29

PART D:	GENERAL	30
25.	METHOD OF CONTRACTING	30
25.1	DEEDS	30
25.2	OTHER WRITTEN CONTRACTS	30
25.3	OTHER OBLIGATIONS	30
23.3	OTHER OBLIGATIONS	
26.	ACCOUNTS	30
26.1	ACCOUNTS TO BE KEPT	30
26.2	APPOINTMENT OF AUDITORS	30
27.	LIQUIDATION	30
27.1	DISTRIBUTION OF ASSETS	30
PROCEE	DINGS AT MEETINGS OF SHAREHOLDERS	32
4	CHAIDDEDSON	20
1.		<u>32</u> 32
1.1	CHAIRPERSON TO BE CHAIRPERSON OF THE BOARD	
1.2	APPOINTMENT BY DIRECTORS	32
1.3	APPOINTMENT BY SHAREHOLDERS	32
2.	NOTICE OF MEETINGS	32
2.1	NOTICE IN WRITING	32
2.2	CONTENTS OF NOTICE	32
2.3	IRREGULARITIES IN NOTICE	33
2.4	ACCIDENTAL OMISSION TO GIVE NOTICE	33
3.	PROCEEDINGS AT MEETINGS	33
3.1	ADJOURNMENT BY CHAIRPERSON	33
3.2	ADJOURNMENT OR DISSOLUTION OF MEETING IF DISORDERLY	33
3.3	COMPLETION OF UNFINISHED BUSINESS IF MEETING DISSOLVED	33
3.4	OTHER PROCEEDINGS	33
4.	QUORUM	34
4.1	NECESSITY FOR QUORUM	34
4.2	NUMBERS FOR QUORUM	34
7.2		
5.	VOTING	34
5.1	NUMBER OF VOTES	34
5.2	VOTING METHOD	34
5.3	VOTING METHOD - AUDIO, AUDIO/VISUAL	34
5.4	EVIDENCE THAT RESOLUTION CARRIED	34
5.5	WHO MAY DEMAND POLL	35
5.6	WHEN POLL MAY BE DEMANDED	35
5.7	COUNTING OF VOTES	35
5.8	EQUALITY OF VOTES	36
5.9	PROXY HOLDER MAY DEMAND POLL	36

5.10	VOTING ENTITLEMENT	36
5.11	CHAIRPERSON MAY DEMAND POLL	36
5.12	WITHDRAWAL OF DEMAND	36
5.13	POLL TO BE TAKEN AS CHAIRPERSON DIRECTS	36
5.14	POLL ON ELECTION OF CHAIRPERSON	36

6.	PROXIES	36
6.1	RIGHT TO VOTE BY PROXY	36
6.2	RIGHT OF PROXY TO ATTEND	36
6.3	APPOINTMENT OF PROXY	37
6.4	PROXY FORM TO BE SENT WITH NOTICE OF MEETING	37
6.5	Two Way Voting	37
6.6	PROXY FORM MUST NOT NAME PROXY	37
6.7	VALIDITY OF VOTE	37
6.8	DEPOSIT OF PROXY	37

7.	POSTAL VOTES	37
7.1	SHAREHOLDERS MAY NOT EXERCISE THE RIGHT TO VOTE AT A MEETING BY CASTING	А
POSTAL V	OTE UNLESS THE BOARD ELECTS TO THE CONTRARY IN RELATION TO A PARTICULAR	
MEETING	. A POSTAL VOTE MAY BE CAST USING ELECTRONIC MEANS PERMITTED BY THE BOARD	. 37

8.	MINUTES	37
8.1	MINUTES MUST BE KEPT	37
8.2	EVIDENCE	38

9.	SHAREHOLDER PROPOSALS	38
9.1	NOTICE OF MATTER FOR DISCUSSION OR RESOLUTION	38
9.2	NOTICE OF SHAREHOLDER PROPOSAL AT COMPANY'S EXPENSE	38
9.3	NOTICE OF SHAREHOLDER PROPOSAL AT SHAREHOLDER'S EXPENSES	38
9.4	NOTICE OF LATE SHAREHOLDER PROPOSAL TO BE GIVEN IF PRACTICABLE	38
9.5	PROPOSING SHAREHOLDER'S WRITTEN STATEMENT	38
9.6	LIMITS ON OBLIGATION TO INCLUDE STATEMENT	38
9.7	PAYMENT BY SHAREHOLDER OF COSTS	38

<u>10.</u>	CORPORATIONS MAY ACT BY REPRESENTATIVES	39
11.	VOTES OF JOINT HOLDERS	39
12.	LOSS OF VOTING RIGHTS	39
12.1	NO VOTE WHEN AMOUNT OWING ON SHARE	39
12.2	VOTING RESTRICTIONS	39
12.3	DEADLINE FOR CHALLENGE	39
	EEDINGS OF THE BOARD	40
1.	CHAIRPERSON	40
1.1	ELECTION OF CHAIRPERSON	40
1.2	TERMS OF OFFICE	40
1.3	ELECTION OF CHAIRPERSON FOR PARTICULAR MEETINGS	40
2.	NOTICE OF MEETING	40
2.1	CONVENING MEETINGS	40

2.2	NOTICE	40
2.3	IRREGULARITY IN NOTICE	41
0		
3.	METHOD OF HOLDING MEETINGS	<u>41</u> 41
3.1	A MEETING OF THE BOARD MAY BE HELD EITHER:	41
4.	QUORUM	41
4.1	NUMBER CONSTITUTING QUORUM	41
4.2	NO BUSINESS WITHOUT QUORUM	41
4.3	ALTERNATE DIRECTOR MAY BE INCLUDED	41
4.4	INTERESTED DIRECTOR	41
5.	VOTING	41
5.1	NUMBER OF VOTES	41
5.2	CHAIRPERSON HAS CASTING VOTE	42
5.3	MAJORITY	42
5.4	PRESUMPTION AS TO VOTING	42
5.5	DISCLOSURE OF INTERESTS	42
5.6	PERSONAL INVOLVEMENT OF DIRECTORS	42
5.7	INTERESTED DIRECTORS MAY NOT VOTE	42
5.8	EXCEPTION TO VOTING PROHIBITION	43
5.9	ALTERNATE DIRECTOR MAY VOTE	43
6.	MINUTES	43
7.	RESOLUTIONS	43
7.1	WRITTEN RESOLUTION	43
7.2	FORMS OF RESOLUTION	43
7.3	RESOLUTION TO BE KEPT IN MINUTE BOOK	43
8.	NO NOTICE TO DIRECTORS OUTSIDE NEW ZEALAND	43
9.	COMMITTEES	43
<u>10.</u>	OTHER PROCEEDINGS	44

Deleted: PART A: .. INTRODUCTION .. 1 .

[...[2]]

# Constitution of Pacific Edge Limited

Deleted: Biotechnology

	DEFINITIONS AND IN	ITERPRETATION			
	Defined Terms				
	In this Constitution, unless the context otherwise requires:				Deleted: "Aggregate Net in relation to assets has the meaning given to the
	,,,			$\sim$	Deleted: as his or her alternate
	"Act"	means the Companies Act 1993.	- 17	$l \geq$	Deleted: 23
			M		Deleted: "Appraisal Report" has the meaning given to that term by Listing Rule 1.2 (
	"Alternate Director"	means a person appointed by a Director under clause <u>21 to act</u>	J - 18	$\sim$	Deleted: Capitalisation"
	. <b>x</b>	in the place of that Director.		t t t	Deleted: "Bank" means a registered bank in terms of the Reserve Bank of New Zealand Act 1989, a bank having recognition comparable to that of a registered bank under the law of Australia, the United States of America, Japan or the United Kingdom, or any other financial institution approved by the Exchange.
	"Board"	means those Directors who number not less than the required guorum acting together as the board of directors of the			Deleted: "Business Day" means a day on which the Exchange is open for trading.
		Company.		i i	Deleted: "Class" - means a Class of Securities having identical rights, privileges, limitations and conditions and includes or excludes Securities which the Exchange in its discretion deems to be of or not of that Class.
	<b>v</b>		1	/1	Deleted: Biotechnology
	"Company" "Constitution"	means Pacific Edge Limited. means this constitution, as amended from time to time.		t s	Deleted: "Convert". in respect of a Security, means to convert that Security into, or exchange that Security for, a Security of a different sort, whether at the option of the holder, or of the Company, or otherwise, or to subscribe for or obtain a Security of a different sort pursuant to a right conferred by the first mentioned Security and "Conversion" and "Convertible" have corresponding meanings.
	"Director" "Distribution Right"	means a person appointed as a director of the Company in accordance with this constitution, means a right in the nature of:		ii C F a C C L L	Deleted: "Debt Security" - means a Security having any interest in or right to be paid money that is, or is to be, deposited with, lent to, or otherwise owing by, any person (whether or not the interest or right is secured by a charge over any property) and includes a debenture, debenture stock, bond, note or certificate of deposit and any other Security which the Exchange in its sole discretion declares by a Ruling to be a Debt Security but does not include any Security that the Exchange in its sole discretion declares by a Ruling not to be a Debt Security.
	U U	<ul> <li>a. a present or future right to participate in the assets of the Company after payment of claims payable under section 313(1) of the Act; or</li> <li>b. a present or future right to participate in the income or profits of the Company.</li> </ul>			Deleted: "Disqualifying , has the meaning set out in Listing Rule 1.1.2.
					Deleted: in relation to the Company any
			/	11	Deleted: occupying the position of
				~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	Deleted: , by whatever name called
	_			t	Deleted: "Distribution" has the meaning given to that term by the Act.
	v			t	Deleted: "Dividend" , has the meaning given to that term by the Act.
	<b>v</b>				<b>Deleted: "Employee"</b> , in relation to the Company has the meaning given to that term by Listing Rule 7.3.6.

1

1

1

1

Υ		Deleted: "Equity Security" - means a Security of the Company which: - 
"Executive Director"	means a Director who is also an employee of the Company.	Deleted: "Exchange" means the NZX
"Independent Director"	means a Director who is not an executive officer of the Company and who has no Disqualifying Relationship.	
		Deleted:"Interest Group" - has the meaning giver that term by the Act
		Deleted: "Interested". in relation to a Director, has meaning set out in section 139 of the Act.
"Listing Rules"	means the "NZX Listing Rules" of NZX_amended from time to time by NZX.	Deleted: - "Investment - has the meaning given in section 38C of the Securities Act - "Issuer" - has the meaning given to that term by th Listing Rules
▼	•	Deleted: "Issuer" , has the meaning given to that te Listing Rules.
Y		Deleted: "Listing" - has the meaning given in Listin Rule 1.1.2 and the terms "listed" and "list" have a corresponding meaning
<u>"NZX"</u>	means NZX Limited, its successors and assigns and, as the context permits, includes any duly authorised delegate of NZX.	Deleted: S
	context permits, includes any duly autionsed delegate of NZX.	Deleted:
•		Deleted: of the Exchange in force
-		Formatted: Indent: Left: 0 cm, Hanging: 5.43 cm, Tab 1.27 cm, Left
		Deleted: "Material - means, for the purposes of clau 20, a transaction or -
"person"	includes an individual, partnership, firm, company, body corporate, corporation, association, organisation, trust, a state	Deleted: "Minimum Holding" - means a parcel or number of Securities as set out in Appendix 2 of the Listing Rules.
	or government or any agency thereof, a municipal, local or	Deleted:
	regional authority, and any other entity or organisation, whether	Formatted: Font:Bold
	incorporated or not (in each case whether or not having a separate legal personality).	Formatted: Indent: Left: 0 cm, Hanging: 5.43 cm, Tat cm, Left + Not at 5.33 cm
"Personal Representative"	means <u>in relation to</u> : a. a deceased individual Shareholder, the executor,	Deleted: "Offering Document" - means any releval Profile or Investment Statement pursuant to which Securities in the Company are either offered or, in case of a Profile, Listed on the Exchange.
	administrator or trustee of the estate of that Shareholder;	Deleted: "Option" - means an option to acquire a Security.
	b. a bankrupt individual Shareholder, the assignee in bankruptcy of that Shareholder; and	<b>Formatted:</b> Indent: Left: 0 cm, Hanging: 1.27 cm, Tat at 1.27 cm
	c. any other individual Shareholder, a person appointed or	Deleted: "Ordinary - means a resolution that is approved by a simple majority -
	deemed to have been appointed to administer property	Deleted: in relation to
	under the Protection of Personal and Property Rights Act	Deleted: in relation to
	1988, a manager appointed or deemed to have been appointed thereunder, and a donee of an enduring power of attorney complying with the Protection of Personal and Property Rights Act 1988.	Deleted: in relation to
<b>v</b>		Deleted: "Profile" - means a document complying w Listing Rule 7.1.3.
		Deleted: "Prospectus", means a prospectus within

1904396 4422040

|

Ì

•	۲			Deleted: "Quotation" - means, in respect of a Class of Securities, the right to quote bids and offers for that Class of Security on the Exchange and "Quote" and "Quoted" have corresponding meanings.
	<b>.</b>			Deleted: "Recognised . has the meaning given to that term by Listing Rule 1.1.2 "Record Date" has the meaning given to that term in Listing Rule 1.1.2.
			$\langle \rangle \langle$	<b>Deleted: "Record Date"</b> has the meaning given to Listing Rule 1.1.2.
	"Representative"	means:	$\left  \right\rangle$	Deleted: "Records" - means the documents required to be kept by the Company under section 189(1) of the Act.
		<ul> <li>a. person appointed as a proxy under paragraph 6 of the First Schedule;</li> <li>b. a Personal Representative; or</li> <li>c. a representative appointed by a corporation under paragraph 10 of the First Schedule.</li> </ul>		Deleted: "Related . has the meaning given in section 2 (3) of the Companies Act . "Related Party"has, for the purposes of clause 20, the meaning given to that term by Listing Rule 9.2.3 "Renounceable" in relation to a Right or offer of Securities means a Right or offer that is transferable by any holder for the time being to another person (whether or not an existing holder of any Securities to which the Right or offer relates)
	۲ ۲	4		Deleted: "Related Party" has, for the purposes of the meaning given to that term by Listing Rule 9.2.3. " "Renounceable" in relation to a Right or offer of Securities means a Right or offer that is transferable by any holder for the time being to another person (whether or not an existing holder of any Securities to which the Right or offer relates).
	"Share" "Shareholder"	means a <u>share</u> issued, or to be issued, by the Company, means a person whose name is entered in the <u>share register of</u> <u>the Company</u> as the holder for the time being of one or more		Deleted: "Renounceable". in relation to a Right or Securities means a Right or offer that is transferable by any holder for the time being to another person (whether or not an existing holder of any Securities to which the Right or offer relates).
		Shares.		Deleted: "Right" . means any right to acquire any Security or benefit of any kind, whether conditional or not, and whether Renounceable or not.
	"Special Resolution"	means a resolution approved by a majority of 75% of the votes of those Shareholders entitled to vote and voting on the resolution.		Deleted: "Ruling" - means any decision or determination by the Exchange as to the meaning or interpretatiq [9] Formatted: Indent: Left: 0 cm, Hanging: 5.33 cm, Tabs: 1.27 cm, Left
				<b>Deleted: "Security"</b> means any interest or right to participate in any capital, assets, earnings, royalti
	"in writing" or	include any means of reproducing words, figures and symbols		Deleted: n Equity Security
	"written"	in a tangible and visible form.		Deleted: (including an Equity Security)
1.2	Interpretation			Deleted:, as the case may require
1.2	Interpretation			Deleted: S
	In this Constitution, unl	ess the context otherwise requires:		Deleted: R
		pressions defined in the Act or the Listing Rules (whether or not ith an initial capital letter) have the same meaning in this		Deleted: "Share Register" - means the share register for the Company required to be kept in accordan( [1] Deleted: "Solvency Test" - has the meaning given to
	Constitution, e	except where this Constitution expressly provides otherwise.		that term by the Act Deleted: or more
	استعاده فاستراحه والأراج	allowed be address and dependentions uplations to practice of the Art		

- b. the index and clause headings, and descriptions relating to sections of the Act, are included for convenience and must be ignored in interpreting this Constitution;
- c. words importing one gender include the other genders;

Deleted: question
Deleted: \_ (...[12])
Deleted: \_ "Working Day" - has the meaning set out in
of the Act. \_

Deleted: V Deleted: V

1904396 4422040

1

- 4 -
- d. the singular includes the plural and vice versa;
- e. a reference to any legislation includes:
  - i. the statute and any regulations, rules or orders made pursuant to that statute and in existence from time to time; and
  - ii. all amendments, re-enactments and substituted legislation; and
  - any statutory instruments, regulations, rules and orders issued under that legislation;
- f. references to clauses and schedules are references to clauses and schedules in this Constitution, unless stated otherwise. References to paragraphs are references to paragraphs in the Schedules to this Constitution;
- a reference to a Listing Rule includes all amendments to and substitutions for that Listing Rule; and
- h. \_\_\_\_\_references to time are New Zealand time.

#### 1.3 Use of electronic means

Where a legal requirement under the Act is reproduced in this Constitution, that legalrequirement may be met, for the purposes of this Constitution, by using electronic means in accordance with Part 4 of the Contract and Commercial Law Act 2017 to meet that legal requirement under the Act. In this clause, the term "legal requirement" has the same meaning given to it by section 219(2) of the Contract and Commercial Law Act 2017.

# 2. APPLICATION OF THE LISTING RULES

# 2.1 Compliance with Listing Rules

- a. While the Company is Listed, the Company <u>must</u> comply with the Listing Rules subject to <u>any enactment or rule of law, and to</u> any <u>waiver or Ruling</u>, If a provision of this Constitution is inconsistent with the Listing Rules, the Listing Rules prevail.
- b. For so long as the Company is Listed, those provisions of the Listing Rules which are required to be contained or incorporated by reference in this Constitution (and as modified by any <u>waiver or</u> Ruling relevant to the Company), will be deemed to be incorporated in this Constitution and have the same effect as if they were set out in full with any necessary modification.

# Effect of Rulings by the <u>NZX</u>

If the <u>NZX</u> has <u>granted a waiver or Ruling</u> in relation to the Company authorising any act or omission, which in the absence of that <u>waiver or Ruling</u> would be in contravention of the Listing Rules or this Constitution, <u>unless a contrary intention appears in this</u> **Deleted:** \_g. to the extent permitted by the Act, if there is a conflict between a provision or definition in the Act and a provision or definition in this Constitution, this Constitution will prevail to the extent permitted by the Act; \_

# Deleted: h Deleted: and Deleted: [...[13]] Formatted: Font:Frutiger LT Std 45 Light Formatted: Font:Not Bold Formatted: Font: Arial Formatted: Font:Bold Formatted: Indent: Left: 0 cm, Hanging: 1.27 cm, Keep with next Formatted: Condensed by 0.15 pt Deleted: EXCHANGE Formatted: Not Highlight Deleted: on the Exchange Deleted: shall Deleted: the Act, applicable legislation and Deleted: of the Exchange in force Deleted: or any Ruling shall Formatted: Not Highlight Deleted: S Deleted:

Deleted: they may be

Formatted: Not Highlight

# Formatted: Highlight

Deleted: <#>Where any modifications made to the Listing Rules by the Exchange would, but for clause 2.1b, require a change in this Constitution, those modifications will be deemed to be incorporated by reference in this Constitution upon the date specified (either as a certain date or the date of expiry of any period of notice) by the Exchange as the date upon which such incorporation by reference is deemed to take effect.

# Deleted: Exchange

Deleted: Exchange Deleted: made

1904396 4422040

2.2

Constitution, that act or omission is deemed to be authorised by the Listing Rules and this Constitution.

#### 2.3 Effect of failure to comply with Listing Rules

Failure to comply with the Listing Rules, or with a provision of the Constitution corresponding with a provision of the Listing Rules, shall not affect the validity or enforceability of any transaction, contract, action or other matter whatsoever (including the proceedings of, or voting at, any meeting) done or entered into by, or affecting, the Company, provided that:

- a party to a transaction or contract who knew of the failure to comply with the a. Listing Rules or the clauses referred to above, as the case may be, is not entitled to enforce that transaction or contract; and
- b. this clause 2.3 shall not Jimit the rights of Equity Security holders, against the Company or the Directors,

#### PART B: SHARES AND SHAREHOLDERS

#### 3. **RIGHTS ATTACHING TO SHARES**

#### **Classes of Shares** 3.1

1

**Deleted:** (whether such provision is set out in full in this Constitution or incorporated in it pursuant to clause 2.1(c))

Deleted: affect
Deleted: any
Deleted: of Securities of the Company
<b>Deleted:</b> arising from failure to comply with the Listing Rules or any of those provisions of this Constitution
Deleted: <#>Obligations prior to Listing

Formatted: NoNum

Without limiting the Classes of Shares or securities that may be issued, Shares in the Deleted:					
Company may be issued <u>on terms that they</u> ;		Formatted: Indent: First line: 0 cm			
are convertible into an evolution acable for Charges		Deleted: , of different Classes with different			
a. <u>are convertible into or exchangeable for Shares;</u>	N.	Deleted: , rights or restrictions of any kind including			
b. are redeemable:		Formatted: Not Expanded by / Condensed by			
· · · · · · · · · · · · · · · · · · ·		Formatted: NoNum, Indent: Left: 0 cm			
c. confer preferential rights to Distributions and capital or income which may be	and the second	Formatted: Not Expanded by / Condensed by			
made subject to the power of the Directors to make Distributions;		Formatted: NoNum, Indent: Left: 0 cm			
d. confer special, limited or conditional voting rights;					
	5	Formatted: Not Expanded by / Condensed by			
e. do not confer voting rights; or	-	Formatted: NoNum, Indent: Left: 0 cm			
f. have limitations or restrictions on transferability.	~	Formatted: Not Expanded by / Condensed by			
	No.	Formatted: NoNum, Indent: Left: 0 cm			
	N	Deleted: ordinary Shares; and			
1		Deleted:			
ISSUE OF SHARES AND OTHER SECURITIES Board Has Power to Issue Shares and Other Securities		Deleted: subject to this Constitution and to any rights previously conferred on the holders of any existing Share or Class of Share			
		Deleted: <#>Terms of Issue of Series A Convertible Preference Shares			
Subject to the Listing Rules, the Act and this Constitution, the Board may issue new		Deleted: clause 4.3			
Shares, Financial Products, or Options to any person and in any number it thinks fit.		Deleted: Securities			
Pre-emptive Rights on Issue		Deleted: that are Convertible into Shares			

1904396 4422040

4 4.1

4.2

The provisions of section 45 of the Act do not apply to the issue of further Shares.

Deleted: <#>Prohibition on Issue <#>Specifically Approved Issues

Deleted: <#>Specifically Approved Issues

Formatted: Indent: Left: 1.27 cm, No bullets or numbering

		Deleted: Pro rata and \$5,000 Offers
	5~	Deleted:
d		Formatted: Heading 4, Indent: Left: 1.27 cm, Hyphenate, Tabs:Not at 0 cm + 2.79 cm + 4.06 cm + 5.43 cm + 6.1 cm + 6.84 cm + 7.56 cm + 7.62 cm
b.		Deleted: Offers not Accepted and Fractional Entitlements: the Board may issue any Equity Securities in respect of which the offer is not accepted, or which because of fractional entitlements are not otherwise offered, to such persons and in such manner as the Board considers equitable and in the interests o the Company if the price and terms and conditions of issue are not materially more favourable to the persons to whom they are issued than the terms of the original offer, and the issue is completed within 3 months after the close of the original offer; and
solidation and Subdivision of Shares Board may:		Deleted: . <#>Issues within 15% Limit . Employee Share Issues . . For the purposes of this clause 4.7: .
<ul> <li>consolidate and divide the Shares or the Shares of any Class in proportion to those Shares or the Shares in that Class; or</li> </ul>		<b>Characteristics</b> (7.11) <b>Formatted:</b> Indent: Hanging: 1.27 cm
<ul> <li>b. subdivide the Shares or the Shares of any Class in proportion to those Shares or the Shares in that Class,</li> </ul>		Deleted: <#>Issues within 15% Limit . Employee Share Issues . For the purposes of this clause 4.7: . <#>Other Permitted Issues .
τ		Deleted: Employee Share Issues For the purposes of this clause 4.7: <pre>#&gt;Other Permitted Issues</pre>
		Deleted: For the purposes of this clause 4.7:
	-	Deleted: <#>Other Permitted Issues
		Deleted: Conversion of Convertible Securities: the issue is made upon Conversion of any Equity Security or any other Security from time to time issued by the Company if the terms of issue of those Securities provided for Conversion to the kind of Securities issue or
		Deleted:
		Deleted: <#>Transactions Deemed to be Issues of Securities

Deleted: <#>Issue of Securities Affecting Control [27]

Deleted: However any such consolidation or subdivision made to the class of ordinary Shares must also b(...[28]) Deleted: <#>Entitlements to Third Party Securities

Deleted: <#>Change of Option on a Bonus Issu [33] Deleted: <#>Change of Option on Consolidation or

[...[29]]

[...[32]]

[...[34]]

Deleted: so that each Shareholder holds a Deleted: ately fewer number of Shares Deleted: so that each Shareholder holds a Deleted: ately greater number of Shares

<#>Notice of Meetings

Underlying Securities

Subdivision

# 5. MODIFICATION OF SHAREHOLDER RIGHTS

#### 5.1 Procedure for Modification of Shareholder Rights

The Company shall comply with the provisions of sections 116 and 117 of the Act before taking action affecting the rights attached to any Shares. For the purposes of this clause, those sections shall be deemed to be modified so that:

- a. reference in those sections to "shares" shall (subject to clause 5.2) be deemed to be references to all Equity Securities, and references to "shareholders" shall be read accordingly; and
- b. in respect of Equity Securities which are not Shares:
  - references to a special resolution shall be deemed to be references to a resolution approved by a majority of 75% of votes of the holders of those <u>Financial Products</u> entitled to vote and voting; and
  - ii. references to the constitution shall be deemed to be references to the document which governs the rights of those Equity Securities.

# 5.2 Exceptions

The Company shall not be required by clause 5.1 to comply with sections 116 and 117 of the Act in respect of actions that affect the rights attached to:

- a. Equity Securities which are not Quoted; or
- b. Equity Securities which are not Shares if those Equity Securities were issued on terms which expressly permitted the action in question to be taken without the approval of holders of those Equity Securities, and those terms were clearly disclosed in the Offer, Document (if any) pursuant to which those Equity Securities were offered.

# 5.3 Issue of Equal Ranking Equity Securities

The issue of further Equity Securities which rank equally with, or in priority to, existing Equity Securities, whether as to voting rights, Distributions or otherwise, is deemed not to be an action affecting the rights attaching to those existing Equity Securities.

# 5.4 Cancellation of Unpaid Amounts

The Company shall not cancel, reduce or defer any obligation to pay an unpaid amount on any Equity Security without the authority of an Ordinary Resolution of the Company.

# 5.5 Actions not invalid

The taking of any action by the Company affecting the rights attached to any Quoted Equity Securities other than Shares shall not be invalid by reason only that the action was not approved in accordance with the provisions of clause 5.1.

1904396 4422040

Deleted: Securities
Deleted: Securities
Deleted: Securities
Deleted: ing
Deleted: ing
Deleted: ing
Deleted: ing
Deleted: any
Deleted: , is subject to clause 4.3

# 6. CALLS ON SHARES

# 6.1 Board may make Calls

- a. The Board may make such calls as the Board thinks fit on the Shareholders in respect of any moneys unpaid on any Shares held by them which are not by the conditions of issue of those Shares made payable at a fixed time or times.
- b. Shareholders shall comply with the terms of payment set out in the Board resolution.
- c. A call may be revoked, reduced or postponed at any time by the Board.

# 6.2 Time of Call

A call is deemed to be made when the resolution of the Board <u>authorising</u> the call is passed, unless that resolution provides otherwise.

# 6.3 Fixed Instalment Deemed Calls

An amount which by the terms of issue of a Share is payable on issue or at a fixed date is deemed for the purposes of this Constitution to be a call duly made and in respect of which notice and particulars of the call have been given and shall be payable on the date on which the amount is due.

# 6.4 Notice of Call

- a. At least 21 days notice of call shall be given to the holder of the Shares in respect of which the call is made, specifying the amount, time and place of payment at the time the call is made.
- b. The Company is not required to give notice and particulars of call to a subsequent holder of those Shares.

# 6.5 Different amounts

The Board may, on the issue of Shares, differentiate between the holders as to the amount of calls to be paid and the times of payment.

#### 6.6 Liability

The joint holders of a Share are jointly and severally liable to pay all calls in respect of those Shares.

# 6.7 Interest and Expenses

If a call is not paid by the day appointed for payment, the person from whom the sum is due shall be liable to pay:

- a. all interest on that sum from the date payment was due to the date of actual payment at such rate as the Board may determine;
- b. all expenses which the Company has or may incur by reason of non-payment of the call.

The Board may waive payment of all or part of that interest or expense.

1904396 4422040

Deleted: making

# 6.8 Proceedings for Recovery of Call

In any proceedings for recovery of a call:

- a. it will be conclusive evidence of the debt if it is proved that:
  - i. the name of the Shareholder sued is entered in the Share Register as the holder, or one of the holders, of the Shares to which the call relates; and
  - ii. either the resolution making the call is entered in the Company's Records and notice of the call has been duly given; or
  - iii. the terms of issue of a Share require the sum to be paid on a certain date; and
- b. it is not necessary to prove the appointment or qualification of any member of the Board which made the call nor any other matter.

# 6.9 Payment in Advance of Calls

The Company may if it thinks fit, receive from any Shareholder in advance any amount uncalled and unpaid upon any Shares held by that Shareholder and may, until the date on which the amount becomes payable pursuant to a call, pay interest on the amount at such rate as the Board and the Shareholder agree. The Company may at any time repay the amount so advanced.

# 7. LIEN ON SHARES

# 7.1 Lien on Shares

The Company has a first and paramount lien upon each Share registered in the name of a Shareholder (whether solely or jointly with others), the proceeds of sale of the Share, and all Distributions made in respect of the Share for:

- a. all unpaid calls owing in respect of the Share and interest owing under clause 6.7 (if any); and
- b. such amount as the Company may be called upon to pay under any legislation in respect of the Share, whether or not the due date for payment has arrived.

# 7.2 Waiver of Lien

Unless otherwise agreed between the Company and the relevant Shareholder, the registration of a Transfer of a Share shall operate as a waiver of any lien which the Company may have on that Share, except as provided in clause 14.2.

# 8. FORFEITURE OF SHARES

# 8.1 Notice Requiring Payment of call

If a Shareholder fails to pay any call or instalment of a call the Company may at any time thereafter by notice to that Shareholder require payment of the amount unpaid together with any accrued interest and all expenses incurred by the Company by reason of such non-payment.

# 8.2 Contents of Notice

The notice referred to in clause 8.1 shall specify a further date (not earlier than 14 days after the date of service of the notice) on or before which the payment is to be made and shall state that, if payment is not made by the specified date, the Share in respect of which the call or instalment of a call is due, is liable to be forfeited.

# 8.3 Forfeiture for Non-Payment

If payment is not made by the date specified in the notice then, at any time thereafter before the payment required by the notice has been made, any Share in respect of which the notice has been given may be forfeited by a resolution of the Board. The forfeiture shall include all Dividends declared in respect of the forfeited Share and not paid before the forfeiture.

# 8.4 **Evidence** of Forfeiture

A statutory declaration by a Director or any other person authorised by the Board that a Share has been forfeited on a specified date will be conclusive evidence of that forfeiture.

# 8.5 Cancellation of Forfeiture

A forfeiture may be cancelled at any time before the sale of the forfeited Share, on such terms as the Board thinks fit.

# 8.6 Effect of Forfeiture

The holder of a Share which has been forfeited ceases to be a Shareholder in respect of the forfeited Share, but remains liable to the Company for all money payable in respect of the forfeited Share.

# 9. SALE OF SHARES SUBJECT TO LIEN OR FORFEITURE

# 9.1 Power of Sale

The Company may sell, in such manner as the Board thinks fit, any forfeited Share or any Share on which the Company has a lien. No sale may be made unless:

- a. the sum in respect of which a lien exists is due and payable; and
- b. until 14 days have elapsed after a notice which requires payment of the amount owing has been given to the Shareholder entitled to receive notice of meetings of Shareholders in respect of the Share.

1904396 4422040

# Deleted: Notice

Deleted: When a Share has been forfeited, the Company shall give notice of the resolution to the Shareholder in whose name the Share stood immediately prior to the forfeiture, and shall enter in the Share Register details of the forfeiture. Deleted: n entry in the Share Register shall

# Deleted: e fact and the date of

 $\ensuremath{\textbf{Deleted:}}$  against all persons claiming the entitlement to the Share

# 9.2 Application of Proceeds of Sale

I

I

If any Shares are forfeited and sold or are sold to enforce a lien, any residue after the satisfaction of unpaid calls, instalments, premiums or other amounts and interest thereon, and expenses, shall be paid to the previous holder, or to the executors, administrators or assigns of the previous holder.

			 Deleted:
9.3	Sale	Procedure	 Deleted:
	a.	In order to give effect to any sale in the exercise of the powers given to it under clause 9.1 the Board may authorise any person to execute any relevant documentation,	 Deleted: Transfer of the Shares to the purchaser
	b.	The purchaser will be registered as the Shareholder of the Shares which are transferred, and will not be bound to see to the application of the purchase money.	
	c.	The purchaser's title to the Shares will not be affected by any irregularity or	 Deleted: e
		invalidity in the forfeiture, sale or other disposal of the Shares.	 Deleted: on
	d.	The remedy of any person aggrieved by the sale will be in damages only and against the Company exclusively.	
	e.	If the certificate for the Shares is not delivered to the Company the Board may issue a new certificate distinguishing it as the Board thinks fit from the certificate not delivered.	
	f.	The registration of a <u>share transfer in respect</u> of the Shares shall operate as a waiver of the lien by the Company in respect of those Shares, but not as a release of any outstanding liability owed by any previous Shareholder.	 Deleted: T
).	DIST	RIBUTIONS TO SHAREHOLDERS	
D.1	Satis	faction of Solvency Test	
	after t this C	Board, if it is satisfied on reasonable grounds that the Company will, immediately the Distribution, satisfy the Solvency Test, may subject to section 53 of the Act and Constitution, authorise a Distribution at a time, and of an amount, and to any sholders it determines and otherwise in accordance with the terms of issue.	
0.2	Rest	riction on Certain Distributions	
	The E	Board shall not authorise a Dividend:	
	a.	in respect of some but not all the Shares in a Class; or	
	b.	that is of a greater value per Share in respect of some Shares of a Class than it is in respect of other Shares of that Class;	
	the a this C that S	s the amount of the Dividend in respect of a Share of that Class is in proportion to mount paid to the Company in satisfaction of the liability of the Shareholder under constitution or under the terms of issue of the Share. A Shareholder may waive Shareholder's entitlement to receive a Dividend by written notice to the Company d by or on behalf of the Shareholder.	

# 10.3 Method of Payment

- a. Any Distribution, interest, or other money payable in cash in respect of Shares may be paid as determined by the Board by automatic payment to a bank account nominated in writing by the Shareholder or by cheque sent through the post to the registered address of the Shareholder, or in the case of joint Shareholders, to the registered address of that one of the joint Shareholders who is first named in the Share Register or to such person and to such bank account or address as the Shareholder or joint Shareholders may in writing direct.
- b. The Company shall not be responsible for any loss arising from any mode of transmission referred to in this clause 10.3. Every such cheque must be made payable to the order of the person to whom it is sent. Any one of two or more joint Shareholders may give effectual receipts for any Dividends, bonuses, or other money payable in respect of Shares held by them as joint Shareholders.

# 10.4 Shares in Lieu of Dividends

The Board may issue Shares to any Shareholders who have agreed to accept the issue of Shares, wholly or partly, in lieu of a proposed Dividend or proposed future Dividends if:

- a. the right to receive Shares, wholly or partly, in lieu of the proposed Dividend or proposed future Dividends has been offered to all Shareholders of the same Class on the same terms; and
- b. if all Shareholders elected to receive the Shares in lieu of the proposed Dividend, relative voting or Distribution Rights, or both, would be maintained; and
- c. the Shareholders to whom the right is offered are afforded a reasonable opportunity of accepting it; and
- d. the Shares issued to each Shareholder are issued on the same terms and subject to the same rights as the Shares issued to all Shareholders in that Class who agree to receive the Shares; and
- e. the provisions of section 47 of the Act are complied with by the Board.

The Board may vary, suspend or terminate any plan to issue Shares in lieu of Dividends.

#### 10.5 Bonus Issues

The Board may resolve to apply any amount which is available for Distribution either:

- a. in paying up in full Shares or other <u>Financial Products of the Company to be</u> issued credited as fully paid to:
  - i. the Shareholders who would be entitled to that amount if it were distributed by way of Dividend, and in the same proportions; and

1904396 4422040

Deleted: Subject to clause 4.5.b, the

Deleted: Securities

- ii. if applicable, the holders of any other <u>Financial Products of the</u> Company who are entitled by the terms of issue of such <u>Financial</u> <u>Products to participate in bonus issues by the Company, whether at</u> the time the bonus issue is made to the Shareholders, or at some later time, in accordance with their respective entitlements; or
- b. in paying up any amount which is unpaid on any Shares held by the Shareholders referred to in subclause (a)(i); or

partly in one way and partly in the other.

# **10.6** Shareholder Discounts

The Board may resolve that the Company offer Shareholders discounts in respect of some or all of the goods sold or services provided by the Company (including any subsidiary of the Company) if:

- a. the Board has previously resolved that the proposed discounts are:
  - i. fair and reasonable to the Company and to all Shareholders; and
  - ii. to be available to all Shareholders or all Shareholders of the same Class on the same terms; and
- the Board is satisfied on reasonable grounds that the Company at the time of the offering of the discount scheme or its continuation satisfies the Solvency Test.

# 10.7 Deduction

The Board may deduct from a Distribution payable to a Shareholder any amount which is due and payable by the Shareholder to the Company on account of:

- a. unpaid calls or otherwise in relation to any Shares held by that Shareholder; and
- b. such amounts as the Company may be called upon by law to pay in respect of those Shares, including withholding and other taxes.

#### 10.8 No Interest

No Distribution bears interest against the Company <u>unless the applicable terms of issue</u> of Shares expressly provide otherwise.

# 10.9 Unclaimed Distributions

- a. All Distributions unclaimed for 1 year after having been authorised may be invested or otherwise made use of by the Board for the benefit of the Company until claimed. The Company shall not be required to set those monies apart from its other funds or to regard it as being held on trust. All Distributions unclaimed for 5 years after having been declared may be forfeited by the Board for the benefit of the Company.
- b. The Board must, however, annul any such forfeiture and agree to pay a claimant who produces, to the Board's satisfaction, evidence of entitlement to

1904396 4422040

Deleted: Securities

Deleted: Securities

the amount due to such claimant, unless in the opinion of the Board such payment would embarrass the Company.

# 11. ACQUISITION AND REDEMPTION OF COMPANY'S OWN SHARES

# 11.1 Company may Purchase, Acquire or Redeem its own Shares

Subject to the Act, the Listing Rules and this Constitution, the Company may:

- a. purchase or otherwise acquire Shares or other Equity Securities from one or more Shareholders; and
- b. redeem any redeemable Shares or other Equity Securities; and
- c. hold any Shares or other Equity Securities so purchased, acquired or redeemed,

# 12. FINANCIAL ASSISTANCE BY THE COMPANY TO PURCHASE ITS SHARES

#### 12.1 Financial Assistance

The Company may give financial assistance for the purpose of, or in connection with, the acquisition of Equity Securities issued, or to be issued, by the Company only if the giving of that assistance complies with <u>Act and this Constitution, and is permitted by the Rules</u>.

# 12.2 Permitted Financial Assistance

The Company may only give financial assistance of the nature referred to in clause 12.1 if:

- a. the financial assistance is not given in whole or in part to any Director, Associated Person of a Director, or Employee, and the amount of the financial assistance, together with the amount of all other financial assistance given by the Company under this subclause during the <u>12</u> months preceding the date of giving of the financial assistance does not exceed 5% of the Average Market Capitalisation of the Company; or
- the financial assistance is given to Employees and the amount of the financial assistance, together with the amount of all other financial assistance given by the Company under this subclause b. during the;
  - 12 months preceding the date of giving of the financial assistance, does not exceed 2% of the Average Market Capitalisation of the Company; and
  - five years preceding the date of giving of the financial assistance does not exceed 5% of the Average Market Capitalisation of the Company; and

the financial assistance is not given to any Director or Associated Person of a Director unless it is given to a Director, or an Associated Person of a

ne or	
d or	<b>Formatted:</b> Heading 4, Indent: Left: 1.38 cm, Hanging: 1.16 cm, Widow/Orphan control, Hyphenate, Tabs:Not at 0 cm + 1.27 cm + 2.79 cm + 4.06 cm + 5.43 cm + 6.1 cm + 6.84 cm + 7.56 cm + 7.62 cm
	Deleted: ; . <#>Prohibitions on Purchase or Acquisition . <#>Acquisition or Redemption with Approval of Equity Security Holders
	Deleted: <#>Prohibitions on Purchase or Acquisition <#>Acquisition or Redemption with Approval of Equity Security Holders
with, if the	Deleted: <#>Acquisition or Redemption with Equity Security Holders

Deleted: clauses 11.2 and 11.3

**Deleted:** clause 12.2 or is approved in accordance with clause 12.3

Deleted: shorter of the period of

**Deleted:** and the period from the date on which the Company was Listed on the Exchange to the date of the giving of financial assistance

Deleted: shorter of the period of

**Deleted:** and the period from the date on which the Company was Listed on the Exchange to the date of giving of the financial assistance

**Deleted:** and the period from the date on which the Company was Listed on the Exchange to the date of giving of financial assistance

Director, solely in that person's capacity as a trustee of a bona fide employee share scheme, superannuation scheme, or the like, in which that Director or Associated Person has no beneficial interest; or

c. the financial assistance is offered or given so that all holders of Equity Securities are treated, or given the opportunity to be treated, on the same basis.

# 12.3 Financial Assistance with Approval of Equity Security Holders

The Company may give financial assistance of the nature referred to in clause 12.1 if the precise terms and conditions of the giving of that financial assistance have been approved by separate resolutions (passed by a simple majority of Votes) of the members of each separate group whose rights or entitlements are materially affected in a similar way by the giving of the financial assistance and the giving of the financial assistance is given within 6 months after the passing of the relevant resolutions.

# 13. TRANSFER OF SHARES

# 13.1 Transferability

Subject to any restrictions, contained in this Constitution and to any applicable Deleted: the legislation, a Share is freely transferable.

# 13.2 Form of Transfer

A Shareholder may transfer all or any of its Shares in accordance with the Act, by using a wholly or partly electronic system for the transfer of securities which has been approved by any statute or regulations in New Zealand where the transfer takes place in New Zealand,

# 13.3 Transfers Executed Overseas

Where an instrument of transfer would have complied with the requirements of the Act if it had been executed in New Zealand, it may nevertheless be registered by the Company if it is executed under the common seal of a corporation as transferor or if the signature of the transferor has been witnessed by a person who has added his or her occupation and address after his or her signature.

# 13.4 Other Forms of Transfer

Every instrument of transfer of Shares not falling within the provisions of clauses 13.2 or 13.3 must:

- a. be signed or executed by or on behalf of the transferor in any manner permitted by law; and
- b. if registration of the transfer imposes any liability on the transferee, the transfer must be signed or executed by or on behalf of the transferee.

# 13.5 Delivery to Company

Deleted: terms

Deleted: T

	<b>Deleted:</b> held by that person by using a form of transfer complying with the Securities Transfer Act 1991 in respect of any Shares disposed of by an "authorised transaction" or a "stock exchange transaction" within the meaning of the Securities Transfer Act 1991
	Deleted: T
	Deleted: provisions of the Securities Transfer Act 1991
	Deleted: T
••••	Deleted: T
	Deleted: T
	Deleted: T
	Deleted: T
	Deleted: T

l	Deleteu.	
1	Deleted: T	

An instrument of transfer, together with the Share certificate (if any) relating to the Shares to be transferred, must be delivered to the Company or to the agent of the Company who maintains the Share Register.

- 17 -

# 13.6 Power to Refuse or Delay Registration

The Board may refuse to register a transfer of any Share if:

- a. the Company has a lien on the Share; or
- the instrument of transfer is not accompanied by the relevant Share certificate (if any) and such other evidence as the Board reasonably requires to prove the title of the transferor to, or right of the transferor to transfer, the Share; or
- c. registration of the transfer (together with registration of any further transfer or transfers then held by the Company and awaiting registration) would result in the proposed transferee holding Shares of less than the Minimum Holding; or
- d. such action is permitted by any legislation or the Listing Rules,

provided that the Board resolves to exercise its power under this clause within 30 <u>Business</u> Days after receipt of the relevant transfer and notice of the resolution is sent to the transferor and to the transferee within 5 <u>Business</u> Days of the resolution being passed by the Board.

# 13.7 Compulsory Disposal When Holding Less than Minimum Holding

The <u>Company</u> may at any time give notice to a Shareholder holding less than a Minimum Holding that it intends to exercise its rights under this clause to dispose of the Shares unless within three months (or such longer period as the Board may determine) the relevant Shares are disposed of or further Shares are acquired by the Shareholder to bring the relevant holding up to a Minimum Holding. If, after such period, the Shares then registered in the name of the Shareholder are less than the Minimum Holding.

- a. The Company may sell the Shares by means of a transaction on the Main Board (including through a broker acting on the Company's behalf).
- b. To give effect to any sale under clause 13.7a, the Board may authorise the transfer the Shares sold under this clause to a purchaser of the Shares through the Main Board or in some other manner approved by NZX, at the best price reasonably obtainable at the relevant time.
- c. The Shareholder is deemed to have authorised the Board to act on behalf of the Shareholder in relation to the sale of the relevant Shares, and to sign all documents relating to such sale.
- d. The proceeds of sale of any Shares sold under this clause must be applied as follows:
  - i. first, in payment of any reasonable sale expenses;

ii. second, in satisfaction of any unpaid calls or any other amounts owing to the Company in respect of the Shares, and any interest payable on such amounts; and Deleted: T

Deleted: T

	Deleted: T
-	Deleted: T
	Deleted: T
	Deleted: T
	Deleted: T
$\mathbb{N}$	Deleted: T
N	Deleted: T
Ň	Deleted: a
	<b>Deleted:</b> of Shares of the relevant Class standing in the name of the Transferee
Ì	Deleted: and a Ruling;
	Deleted: Working
Ĵ	Deleted: T
Ń	Deleted: T
V	Deleted: T
	Deleted: Working
	Deleted: Board
Ń	Deleted: three months
Q	Deleted: any
	Deleted: of Shares of any Class
	<b>Deleted:</b> the Board is not satisfied that suitable arrangements have been made for the disposition of the relevant Shares, or the acquisition of further Shares
	Formatted: Not Expanded by / Condensed by
	Formatted: NoNum, Indent: Left: 0 cm, First line: 0 cm
	Deleted: a
Ĵ	Deleted: arrange for the sale of
Ń	Deleted: the
Ń	Deleted: relevant
N	Deleted: on behalf of the Shareholder,
V	Deleted: Exchange
M	Deleted: ; and
N	Deleted: b
N	Deleted: t
V	Deleted: shall be
$\langle \rangle$	<b>Deleted:</b> which may be required to give effect thereto; and
	Deleted: C
V	Deleted: t
	Deleted: net

- iii. the residue, if any, must be paid to the person who was the shareholder immediately before the sale or his or her executors, administrators or assigns,
- e. The purchaser is not bound to see to the application of the purchase money, nor shall the title to the Shares be affected by any irregularity or invalidity in the procedures under this Constitution relating to the sale. The remedy of any person aggrieved by the sale is in damages only and against the Company exclusively.

A certificate, signed by a Director that records that a power of sale under this clause hasarisen and is exercisable by the Company is conclusive evidence of the facts stated in that certificate.

# 13.8 When Transfer Effective

A transferor remains the holder of the Share until the name of the transferee is entered in the Share Register in respect of the Share.

# 13.9 Multiple Registers

The Share Register may, by resolution of the Board, be divided into two or more registers, kept in different places.

# 14. TRANSMISSIONS AND EQUITABLE INTERESTS

# 14.1 Transmission on death of Shareholder

If a Shareholder dies the survivor, if the deceased was a joint Shareholder, or the Personal Representative, shall be the only persons recognised by the Company as having any title to or interest in the Shares of the deceased Shareholder but nothing in this clause shall release the estate of a deceased joint Shareholder from any liability in respect of any Share or constitute a release of any lien which the Company may have in respect of any Share.

# 14.2 Rights of Personal Representatives

A Personal Representative of a Shareholder:

- a. is entitled to exercise all rights (including without limitation the rights to receive Distributions, to attend meetings and to vote in person or by Representative), and is subject to all limitations, attached to the Shares held by that Shareholder; and
- b. is entitled to be registered as the holder of those Shares, but such registration shall not operate as a release of any rights (including any lien) to which the Company was entitled prior to registration of the Personal Representative pursuant to this subclause.

# 14.3 Joint Personal Representatives

Where a Share is subject to the control of two or more persons as Personal Representatives they shall, for the purposes of this Constitution, be deemed to be joint holders of the Share.

1904396 4422040

**Deleted:** shall be held on trust by the Company for, and paid (together with interest at such rate as the Board deems appropriate) to, the Shareholder, on surrender of the certificate (if any) relating to the relevant Shares

Formatted: Font:Arial

Deleted: No purchaser or other person dealing with the Company shall be concerned to enquire whether the power of sale specified in this clause has become properly exercisable, or as to the propriety or regularity of a sale made in purported exercise of that power, or as to the application of the proceeds of sale received by the Company. The receipt of the Company is a good discharge to the purchaser for the purchase price, and no question may be raised as to the title of the purchaser to Shares sold in purported exercise of the power of sale specified in this clause.

Formatted: Normal, Indent: Left: 0 cm, Hanging: 1.27 cm, Space After: 0 pt, Don't hyphenate, Tabs: 0 cm, Left + 1.27 cm, Left + 2.79 cm, Left + 4.06 cm, Left + 5.43 cm, Left + 6.1 cm, Left + 6.84 cm, Left + 7.56 cm, Left + 7.62 cm, Left + Not at 1.25 cm

Formatted: Font:Arial, Condensed by 0.15 pt
Deleted: T
Deleted: T

[...[39]]

Deleted: <#>Transfer of Rights

# 14.4 No Notice of Trusts

No notice of a trust, whether express, implied, or constructive, may be entered on the Share Register.

# 14.5 No Recognition of Equitable Interests

Except as required by law or by this Constitution, no person shall be recognised by the Company as holding any Share upon trust and the Company shall not be bound by, nor be compelled to recognise (even after notice), any equitable, contingent, future or partial interest in any Share, or any interest in any fraction or part of a Share or (except as provided by this Constitution or by law) any other rights in respect of any Share, except an absolute right of the registered holder to the entire Share.

#### 15. SEPARATE PARCELS OF SHARES

# 15.1 Separate Parcels

The Share Registrar of the Company, on request by a Shareholder, or proposed transferee, may register a shareholding in separate parcels identified by a distinguishing word, number or other parcel differentiator. Where a Shareholder's shareholding is so registered, the Company may communicate separately with the Shareholder in respect of each parcel, make Distributions and otherwise act, so far as the Directors consider convenient, as if the separate parcels belonged to different Shareholders.

16.	SHAREHOLDER, REPORTS, INFORMATION AND NOTICES	Deleted: S
16.1	Company Reports and Financial Statements	
	Shareholders of all Classes shall be entitled to attend general meetings and to receive	 Deleted: Equity Security
	copies of all notices, reports and financial statements issued generally by the Company	Deleted:
	to holders of Financial Products carrying Votes.	 Deleted: Securities

# 16.2 Inspection by Shareholders

No Shareholder who is not also a Director is entitled to inspect any accounting or other Records of the Company except as expressly authorised by law or permitted by the Board. Subject to the provisions of section 216 of the Act the Board may from time to time determine whether, to what extent, at what times and places, and under what conditions, the accounting or other Records of the Company are open to inspection.

# 16.3 Method of Service of Notices, Reports and Other Documents

All notices, reports, accounts and other documents required to be sent:

- a. to a Shareholder, shall be sent in the manner provided in section 391 of the Act;
- b. to a holder of any other Equity Security, shall be sent in the same manner, as though that holder were a Shareholder.

# 16.4 Service of notices overseas

1904396 4422040

Deleted: <#>Inspection of Company Records .....[40]

If the holder of a Share or other Equity Security has no registered address within New Zealand and has not supplied to the Company an address within New Zealand for the giving of notices, but has supplied an address outside New Zealand, then notices for that Shareholder shall be posted to that holder at such international address and shall be deemed to have been received by that Shareholder 24 hours after the time of posting.

#### 16.5 Accidental omissions

The failure to send an annual report, notice, or other document to a Shareholder or other Equity Security holder in accordance with the Act or this Constitution does not invalidate the proceedings at a meeting of Shareholders if the failure to do so was accidental.

# 16.6 Joint Shareholders

A notice may be given by the Company to the joint holders of an Equity Security by giving the notice to the joint holder named first in the register in respect of that Equity Security.

# 17. MEETINGS OF SHAREHOLDERS

# 17.1 Annual meetings

The Company shall hold annual meetings of Shareholders in accordance with section 120 of the Act.

# 17.2 Time and place of annual meeting

Each annual meeting shall be held at such time and place as the Board appoints.

# 17.3 Special meetings

All meetings of Shareholders, other than annual meetings, shall be called special meetings.

# 17.4 Calling of special meetings

A special meeting:

- a. may be called at any time by the Board;
- b. must be called by the Board on the written request of Shareholders holding Shares carrying together not less than 5% of the voting rights entitled to be exercised on any of the questions to be considered at the meeting.

# 17.5 Equity Security holders entitled to attend

Equity Security holders of all Classes are entitled to attend meetings of Shareholders.

# 17.6 Methods of Holding Meetings

A meeting of Shareholders may be held either:

- a. by a number of Shareholders, who constitute a quorum, being assembled together at the place, date, and time appointed for the meeting; or
- by means of audio, audio and visual, <u>or electronic</u> communication by which all Shareholders participating and constituting a quorum, can simultaneously hear each other throughout the meeting.

# 18. PROCEDURE AT SHAREHOLDER MEETINGS

1

- 18.1 The provisions of the First Schedule to this Constitution govern proceedings at meetings of Shareholders.
- 18.2 The provisions of the First Schedule to this Constitution, with all necessary modifications, govern proceedings at meetings of Interest Groups and Classes.

Deleted: or

Deleted: <#>RESTRICTION ON DISPOSAL OF ACQUISITION OF ASSETS	ł
Deleted: <#>Restriction - <#>Exception . <#>Notice of Meeting relating to Disposal or Acquisition of Assets .	
Deleted: <#>Exception . <#>Notice of Meeting relating to Disposal or Acquisition of Assets .	[42]
Deleted: <#>Notice of Meeting relating to Dis Acquisition of Assets	posal or
Formatted: Indent: Left: 1.27 cm, No bullets or no	umbering

۷		Deleted: RESTRICTION ON TRANSACTIONS WITH RELATED PARTIES
۲ ۲		Deleted: Restriction . Conditional Agreement Involving a Material Transaction . Permitted Exceptions
Ψ		Deleted: Conditional Agreement Involving a Material Transaction - Permitted Exceptions
۲		Deleted: Permitted Exceptions
Υ		Formatted: Indent: Left: 1.27 cm, No bullets or numbering
PART C: DIRECTORS		<b>Formatted:</b> Normal, Don't hyphenate, Tabs: 0 cm, Left + 1.27 cm, Left + 2.79 cm, Left + 4.06 cm, Left + 5.43 cm, Left + 6.1 cm, Left + 6.84 cm, Left + 7.56 cm, Left + 7.62 cm, Left
9. POWERS OF DIRECTORS		Deleted: The provisions of clause 20.1 do not apply to:
19.1 Management of Company		<#>Notice of Meeting relating to Material Transaction with a Related Party

The business and affairs of the Company shall be managed by, or under the direction or supervision of, the Board.

# 19.2 Exercise of powers by Board

The Board may exercise all the powers of the Company which are not required, either by the Act or this Constitution, to be exercised by the Shareholders.

**Formatted:** Normal, Indent: Left: 0 cm, Hanging: 1.27 cm, Keep with next, Don't hyphenate, Tabs: 0 cm, Left + 1.27 cm, Left + 2.79 cm, Left + 4.06 cm, Left + 5.43 cm, Left + 6.1 cm, Left + 6.84 cm, Left + 7.56 cm, Left + 7.62 cm, Left

Deleted: <#>Notice of Meeting relating to Material Transaction with a Related Party

**Deleted:** The text of any resolution to be put to a meeting for the purposes of clause 20.1 shall be set out in the notice of the relevant meeting. That notice...[44]

# 19.3 Delegation of powers

The Board may delegate to a committee of Directors, a Director, an employee of the Company, or to any other person, any one or more of its powers, other than a power set out in the second schedule to the Act.

# 19.4 Appointment of attorney

The Company may exercise the power conferred by section 181 of the Act to appoint a person as its attorney, either generally or in relation to a specified matter. Any such power of attorney may contain such provisions for the protection of persons dealing with the attorney as the Board thinks fit, and may also authorise any attorney to delegate all or any of the powers, authorities and discretions vested in the attorney.

# 19.5 Ratification by Shareholders

Subject to the provisions of section 177 of the Act the Shareholders, or any other person in whom a power is vested by this Constitution or the Act, may ratify the purported exercise of that power by a Director or the Board in the same manner as the power may be exercised. The purported exercise of a power that is ratified under this clause is deemed to be, and always to have been, a proper and valid exercise of that power.

# 20. APPOINTMENT AND REMOVAL OF DIRECTORS

# 20.1 Number of Directors and Residence

- a. The number of Directors shall not be more than 8,
- b. <u>The Company shall comply with the minimum board composition requirements</u> of the Listing Rules.

#### 20.2 Appointment

Subject to <u>20.1</u>, a person who is not disqualified under the Act may be appointed as a Director at any time by:

- a. an Ordinary Resolution; or
- b. by the Board under clause 20.1 and in accordance with the Listing Rules,

# 20.3 Existing Directors to continue

The persons holding office as Directors at the date of this Constitution continue in office and are deemed to have been appointed as Directors pursuant to this Constitution.

#### 20.4 Nomination of Directors

- a. No person (other than a Director retiring at the meeting) shall be elected as a Director at an annual meeting of Shareholders unless that person has been nominated by a Shareholder entitled to attend and vote at the meeting.
- b. The closing date for nominations shall not be more than two months before the date of the annual meeting at which the election is to take place. The Company shall make an announcement to the market of the closing date for

1904396 4422040

Deleted: nor less than 3

Deleted: At least 2 Directors must be persons who are ordinarily resident in New Zealand. The minimum number of Independent Directors shall be two or, if there are eight or more Directors, three or one-third (rounded down to the nearest whole number of Directors) of the total number of Directors, whichever is the greater.

**Deleted:** Directors may act notwithstanding any vacancy in their body, but if and for so long as their number is reduced below the number fixed by clause 22.1.a as the minimum number of Directors, the continuing Directors may act only for the purpose of increasing the number of Directors to that number or of summoning a general meeting but for no other purposes.

# Deleted: 22.1

**Deleted:** A Director so appointed holds office only until the next annual meeting of the Company but is eligible for re-election at the meeting.

Director nominations and contact details for making nominations no less than <u>10 Business Days</u>, before the closing date for Director nominations.

c. Notice of every valid nomination received by the Company before the closing date for nominations shall be given by the Company to all persons entitled to attend the meeting as part of the notice of the meeting.

# 20.5 Restriction on appointment of several Directors by single resolution

Each resolution of the holders of Equity Securities to appoint a Director must be for the appointment of one Director only. Nothing in this clause prevents the election of two or more Directors by ballot or poll.

# 20.6 Rotation of Directors

A retiring Director continues to hold office: a. until he or she is re-elected; or

b. if he or she is not re-elected, or does not offer himself or herself up for reelection, until the end of the annual meeting or any adjournment of that meeting,

# 20.7 Rotation requirements

Each, Director shall retire from office when required to do so by the Listing Rules, but, subject to the Listing Rules, shall be eligible for re-election (including at any meeting at which the Director retires),

# 20.8 Vacation of Office

A Director ceases to be a Director if he or she:

- a. is removed from office by an Ordinary Resolution or in accordance with the Act or this Constitution; or
- b. dies, or becomes mentally disordered or subject to a property order or personal order made under the Protection of Personal and Property Rights Act 1988; or
- resigns by written notice delivered to the Company at its address for service or at its registered office (such notice to be effective at the time when it is so received unless a later time is specified in the notice); or
- d. retires from office; or
- e, becomes disqualified from being a Director pursuant to the Act; or
- f becomes bankrupt or makes an arrangement or composition with his or her creditors generally; or
- g is an Executive Director and ceases for any reason to be in the salaried employment of the Company or any of its subsidiaries unless the Board decides otherwise; or

	Deleted: 2 months
1	Deleted: together with, or
	Deleted: ,
	<b>Deleted:</b> The Company specify in such notice the Board's view on whether or not the nominee would qualify as an Independent Director.
	Deleted: A
	Deleted: single
	Deleted: for the
//	Deleted: ment of two or more persons
	Deleted: S
	<b>Deleted:</b> s shall not be moved unless a separate resolution that it be so moved has first been approved by the meeting without any Vote being cast against it, but nothing
111	Deleted: "
111	Formatted: Indent: First line: 0 cm
$\mathbb{N}$	<b>Deleted:</b> t the annual meeting of the Company in every year one third of the
]//[	<b>Deleted:</b> s, or if their number is not a multiple of 3 then the number nearest to one third, shall retire from
	Formatted: Indent: Left: 1.27 cm
	Deleted: The Directors to retire shall be those who have been longest in office since their last election or deemed elected. If two of more of those Directors were last elected on the same day, the Directors to retire (unless they otherwise agree) shall be determined by lot. A retiring Director is eligible for re-election, and shall act as a Director throughout the meeting at which that Director retires
	Deleted: Exceptions to
	Deleted: r
	Formatted: Indent: Left: 0 cm, Hanging: 1.52 cm, Keep with next, Keep lines together
	Deleted: In determining the
	<b>Deleted:</b> s who are to retire by rotation at the meeting: b a Director who is retiring pursuant to clause 22.2b shall not be liable to retire by rotation at the meeting if they have offered themselves for re-election and shall not be taken in to account in calculating the number of Directors to retire; .
	Deleted: b a Director who is retiring pursuant to clause shall not be liable to retire by rotation at the meeting if they have offered themselves for re-election and shall not be taken in to account in calculating the number of Directors to retire;
$\sqrt{ }$	<b>Deleted:</b> c. an Executive Director, even though not liable to retire by rotation by virtue of clause 24.2 ( [45])
$\langle \rangle \rangle$	Deleted: <#>Re-election of retiring Director
//	Formatted: NoNum, Indent: Left: 0 cm, First line: 0 cm
	Deleted: d
No.	Deleted: e

Deleted: f

Deleted: @ Deleted: d

h,	has for more than six months been absent without the approval of the Board	
	from meetings of the Board held during that period, unless the Board decides otherwise.	

# 20.9 Appointment by Board

The Board may appoint any person to be a Director either to fill a casual vacancy or as an additional Director. Any Director appointed under this clause <u>20.9</u> will cease to hold office at the commencement of the next annual meeting of the Company.

# 21. ALTERNATE DIRECTORS

# 21.1 Appointment

A Director may from time to time by written notice to the Company appoint any person, who is not already a Director to be that Director's alternate. No Director may appoint an Alternate Director for him or her without the consent of a majority of the other Directors. That appointment may be revoked by a majority of his or her Co-Directors or by the Director having appointed the alternate. No Director may appoint a deputy or agent except by way of appointment of an Alternate Director under this clause <u>21</u>.

#### 21.2 Rights of Alternate Director

Unless otherwise specified by the terms of his or her appointment, an Alternate Director:

- a. is entitled, in the absence or unavailability of the Director who appointed him or her (the **"Appointer**"), to exercise the same rights, powers and privileges (other than the power to appoint an Alternate Director) as the Appointer;
- b. when acting as an Alternate Director is subject to the same duties and obligations as the Appointer;
- c. is not entitled to be given notice of a meeting of the Directors unless the Appointer has give written notice to the Company requesting that notice be given to the Alternate Director;
- d, is entitled to be Indemnified by the Company to the same extent as if he or she were a Director.

# 21.3 Remuneration and expenses

An Alternate Director is not entitled to any remuneration from the Company in his or her capacity as Alternate Director other than out of the remuneration of the Appointer, but is entitled to be reimbursed by the Company for all expenses incurred in attending meetings of the Directors and in the discharge of his or her duties, to the same extent as if he or she were a Director.

# 21.4 Cessation of appointment

An Alternate Director ceases to be an Alternate Director:

a. if the Appointer ceases to be a Director, or revokes the appointment by written notice to the Company; or

1904396 4422040

ecides	
ncy or	
ase to	Deleted: 22.10
oerson, oint an ectors. by the agent	
	Deleted: 23
rector:	
ed him /ileges	
s and	
ss the ice be	
he or	<b>Deleted:</b> d if an alternate for an Executive Director, may not act in that Executive Director's capacity as an

Deleted: g

Employee;

Deleted: e

- 26 -
- on the occurrence of any event that would disqualify the Alternate Director if b. he or she were a Director; or
- if a majority of the other Directors resolve to revoke the Alternate Director's appointment. c.

# Deleted: <#>EXECUTIVE DIRECTOR

[...[47]] Formatted: Heading 3, Indent: Left: 1.27 cm, Hyphenate, Tabs:Not at 0 cm + 1.27 cm + 2.79 cm + 4.06 cm + 5.43 cm + 6.1 cm + 6.84 cm + 7.56 cm + 7.62 cm

# 22. DIRECTOR EXPENSES

Each Director is entitled to be paid for all reasonable travelling, accommodation and other expenses incurred by the Director in connection with the Director's attendance at meetings or otherwise in connection with the business of the Company

#### <#>Not Subject to Retirement by Rotation <#>Remuneration [...[48]] Formatted: Font:Not Bold Deleted: <#>Not Subject to Retirement by Rotation <#>Remuneration [... [49]] Deleted: <#>Remuneration [...[50]] Deleted: S Deleted: REMUNERATION AND OTHER BENEFITS Deleted: <#>Authorisation <#>Interests Register and Certificate <#>Shareholder Approval <#>Power to Increase when Additional Direct(...[51]) Deleted: of Deleted: properly Deleted: them in attending Deleted: of the Board, or any committee of the Board, or meetings of the Shareholders, Formatted: Condensed by 0.15 ptDeleted: ; or Deleted: <#>Interests Register and Certificate <#>Shareholder Approval <#>Power to Increase when Additional Direct(....[52]) Deleted: <#>Interests Register and Certificate <#>Shareholder Approval <#>Power to Increase when Additional Director [53]

Deleted: Powers

Deleted: <#>Shareholder Approval . <#>Power to Increase when Additional Direct(....[54])

Deleted: <#>Power to Increase when Additiona

		Deleted: Notice of proposed increase
23.	INDEMNITY AND INSURANCE	Deleted: Notice of proposed increase .         [
23.1	Indemnity of Directors	
	Subject to clause 23.3, every Director shall be Indemnified by the Company:	Deleted: 26.3
	a. for any costs incurred by him or her in any proceeding that relates for any act or omission in his or her capacity as a Director or a director subsidiary of the Company and in which judgment is given in h favour, or in which he or she is acquitted, or which is discontinued; and	ector of a is or her
	b. in respect of liability to any person other than the Company or company for any act or omission by that Director in his or her capa Director or a director of a subsidiary of the Company, and costs in that Director in defending or settling any claim or proceeding relating such liability.	acity as a curred by
23.2	Other Indemnities	
	Subject to clause <u>23.3</u> , the Company may, with the prior approval of the Indemnify a director of a related company, or an Employee of the Company or company:	

- a. for any costs incurred by him or her in any proceeding that relates to liability for any act or omission by him or her in such capacity and in which judgment is given in his or her favour, or in which he or she is acquitted, or which is discontinued; and
- b. in respect of liability to any person other than the Company or a related company for any act or omission by that director or Employee in such capacity, or costs incurred by that director or Employee in defending or settling any claim or proceeding relating to such liability.

1

1

# 23.3 Exceptions

An Indemnity conferred by clause <u>23.1b</u>, or given pursuant to clause <u>23.2b</u>, shall not apply in respect of:

- a. any criminal liability; or
- in the case of an Employee of the Company or the related company, any liability in respect of a breach of any fiduciary duty owed to the Company or related company; or
- c. in the case of a Director or a director of a related company, any liability in respect of a breach of the duty specified in section 131 of the Act.

# 23.4 Insurance

The Company may, with the prior approval of the Board, Effect Insurance for a Director or Employee of the Company or a director or Employee of a related company, in respect of:

- a. liability, not being criminal liability, for any act or omission by him or her in such capacity; or
- b. costs incurred by him or her in defending or settling any claim or proceeding relating to any such liability; or
- c. costs incurred by him or her in defending any criminal proceedings in which he or she is acquitted.

# 23.5 Definitions

Words given extended meanings by section 162(9) of the Act have those extended meanings in this clause 23,

# 24. PROCEEDINGS OF THE BOARD

24.1 The provisions of the Second Schedule to this Constitution govern the proceedings of the Board. The third schedule to the Act does not apply to the Company.

 Formatted: Indent: Left: 1.27 cm, First line: 0 cm, Keep with next, Tabs: 5.43 cm, Left + Not at 4.83 cm

 Deleted: In this clause 26:

Deleted: 26.1b

Deleted: 26.2b

### PART D: GENERAL

### 25. METHOD OF CONTRACTING

#### 25.1 Deeds

An obligation which, if entered into by a natural person, would, by law, be required to be by deed, may be entered into on behalf of the Company in writing signed under the name of the Company by:

- a. two or more Directors; or
- b. a Director, or other person or persons authorised to do so by the Board whose signature or signatures must be witnessed; or
- c. one or more attorneys appointed by the Company in accordance with section 181 of the Act.

### 25.2 Other written contracts

An obligation or contract which is required by law to be in writing, and any other written obligation or contract which is to be entered into by the Company, may be signed on behalf of the Company by a person acting under the express or implied authority of the Company.

### 25.3 Other obligations

Any other obligation or contract may be entered into on behalf of the Company in writing or orally by a person acting under the express or implied authority of the Company.

### 26. ACCOUNTS

### 26.1 Accounts to be Kept

The Directors shall cause proper books of account to be kept in which shall be kept full true and complete accounts of the affairs and transactions of the Company. The books of account shall be kept at the registered office of the Company, or, subject to the Act, at such other place or places as the Directors shall think fit, and shall always be open to the inspection of any of the Directors.

### 26.2 Appointment of Auditors

Auditors shall be appointed and their duties regulated in accordance with the Act.

### 27. LIQUIDATION

### 27.1 Distribution of assets

If the Company is liquidated the liquidator may, with the approval of Shareholders by Special Resolution and any other sanction required by the Act:

a. divide among the Shareholders in kind the whole or any part of the assets of the Company (whether they consist of property of the same kind or not) and may for that purpose fix such value as the liquidator deems fair in respect of

1904396 | 4422040

Deleted: Subject to clause 30.2, if

b. vest the whole or any part of any such assets in trustees upon such trusts for the benefit of the persons so entitled as the liquidator thinks fit, but so that no Shareholder is compelled to accept any shares or other securities on which there is any liability.

<	Deleted: <#>Preference on Liquidation to Series A Convertible Preference Shares <#>AUDIT COMMITTEE	
	Deleted: <#>AUDIT COMMITTEE	[59] )

1904396 4422040

# **FIRST SCHEDULE**

### **PROCEEDINGS AT MEETINGS OF SHAREHOLDERS**

#### CHAIRPERSON 1.

#### 1.1 Chairperson to be Chairperson of the Board

If the Directors have elected a chairperson of the Board, and the chairperson of the Board is present at a meeting of Shareholders, he or she must chair the meeting, unless or except to the extent that the chairperson considers it not proper or desirable to act as chairperson, either in relation to the entire meeting or to any particular item or items of business to be considered.

#### 1.2 **Appointment by Directors**

If no chairperson of the Board has been elected, or if at any meeting of Shareholders the chairperson of the Board is not present within 15 minutes of the time appointed for the commencement of the meeting or the chairperson considers it not proper or desirable to act as chairperson for all or any part of the meeting, the Directors present may elect one of their number to chair the meeting or that part of the meeting that the chairperson has indicated he or she does not wish to chair.

#### 1.3 Appointment by Shareholders

If at any meeting no Director is willing to act as chairperson, or if no Director is present within 15 minutes of the time appointed for holding the meeting, the Shareholders present must choose one of their number to be chairperson of the meeting.

#### NOTICE OF MEETINGS 2

#### 2.1 Notice in writing

Written notice of the time and place of a meeting of Shareholders must be sent to every Shareholder entitled to receive notice of the meeting and to every Director and to the auditor of the Company not less than 10 Business Days before the meeting, but with the consent of all Shareholders entitled to attend and vote at a meeting, it may be convened by such shorter notice and in such manner as those Shareholders agree.

#### 2.2 Contents of notice

The notice must state:

- the nature of the business to be transacted at the meeting in sufficient detail a. to enable a Shareholder to form a reasoned judgement in relation to it;
- b. the text of any Special Resolution to be submitted to the meeting;
- that a Shareholder entitled to attend and vote at the meeting is entitled to C. appoint a proxy to attend and vote instead of the Shareholder and that a proxy need not be a Shareholder.

So far as is reasonably practicable, resolutions shall be formed in a manner which facilitates two way voting instructions for proxy holders.

1904396 4422040

Deleted: Working

### 2.3 Irregularities in notice

An irregularity in a notice of a meeting is waived if all the Shareholders entitled to attend and vote at the meeting attend the meeting without protest as to the irregularity, or if all such Shareholders agree to the waiver.

### 2.4 Accidental omission to give notice

The accidental omission to give a notice of a meeting to, or the non-receipt of a notice of a meeting by, any person entitled to receive notice does not invalidate the proceedings at that meeting.

### 3. PROCEEDINGS AT MEETINGS

### 3.1 Adjournment By Chairperson

The chairperson may, and if so directed by the meeting must, adjourn the meeting from time to time and from place to place, but no business can be transacted at any adjourned meeting other than the business left unfinished at the meeting from which the adjournment took place. When a meeting is adjourned for 30 days or more, notice of the adjourned meeting must be given as in the case of an original meeting. Except as so provided, it is not necessary to give any notice of an adjournment or of the business to be transacted at an adjourned meeting.

### 3.2 Adjournment or dissolution of meeting if disorderly

If any meeting becomes so unruly, disorderly or inordinately protracted, that in the opinion of the chairperson the business of the meeting cannot be conducted in a proper and orderly manner, the chairperson, notwithstanding any provision to the contrary contained in this Constitution and without the consent of the meeting, may, in his or her sole and absolute discretion and without giving any reason therefor, either adjourn or dissolve the meeting.

### 3.3 Completion of unfinished business if meeting dissolved

If any meeting is dissolved by the chairperson pursuant to paragraph 3.2 of this Schedule, the unfinished business of the meeting shall be dealt with as follows:

- a. in respect of any resolution concerning the approval or authorisation of a Distribution, the Board may, in the exercise of the powers conferred on it by the Act or this Constitution, authorise such Distribution;
- in respect of any resolution concerning the remuneration of the auditors, the meeting shall be deemed to have resolved that the Board be authorised to fix the remuneration of the auditors;
- c. the chairperson may direct that any other item of business which is uncompleted which, in his or her opinion requires to be voted upon, be put to the vote by a poll without further discussion, in accordance with paragraph 5.12 of this Schedule.

### 3.4 Other Proceedings

Except as provided in this Schedule, and subject to the Constitution of the Company, the chairperson may regulate the meetings of Shareholders.

1904396 4422040

## 4. QUORUM

### 4.1 Necessity for quorum

Subject to paragraph 4.3 of this Schedule, no business may be transacted at a meeting of Shareholders if a quorum is not present.

### 4.2 Numbers for quorum

A quorum for a meeting of Shareholders is 5 Shareholders having the right to vote at the meeting, present in person or by proxy or by Representative.

### 5. VOTING

1

#### 5.1 Number of votes

Subject to paragraphs 12.1 and 12.2 of this Schedule and to any rights or restrictions for the time being attached to any Class of Shares:

- a. where voting is by show of hands or by voice every Shareholder present in person or by Representative has one vote;
- b. on a poll every Shareholder present in person or by Representative has:
  - i. in respect of each fully paid ordinary Share held by that Shareholder, one vote;
  - ii. in respect of each Share held by that Shareholder which is not fully paid, a proportion of the vote or votes which would be exercisable if that Share were fully paid equivalent to the proportion of the total issue price of that Share which has been paid (disregarding any payment in advance).

### 5.2 Voting method

In the case of a meeting of Shareholders held under clause 17.6a unless a poll is demanded, voting at the meeting shall be by whichever of the following methods is determined by the chairperson of the meeting:

- a. voting by voice; or
- b. voting by show of hands.

### 5.3 Voting method - audio, audio/visual

In the case of a meeting of Shareholders held under clause 17.6b, unless a poll is demanded, voting at the meeting shall be by the Shareholders signifying individually their assent or dissent by voice.

### 5.4 Evidence that resolution carried

1904396 4422040

 <b>Deleted:</b> ii in respect of each fully paid series A convertible preference Share held by that Shareholder, five votes:

...[60]

Deleted: i

Deleted: <#>No quorum

A declaration by the chairperson of the meeting that a resolution is carried by the requisite majority is conclusive evidence of that fact unless a poll is demanded in accordance with paragraph 5.5 of this Schedule.

### 5.5 Who may demand poll

At a meeting of Shareholders a poll may be demanded by:

- a. not less than 5 Shareholders having the right to vote at the meeting; or
- a Shareholder or Shareholders representing not less than 10 percent of the total voting rights of all Shareholders having the right to vote at the meeting; or
- c. by a Shareholder or Shareholders holding Shares in the Company that confer a right to vote at the meeting and on which the aggregate amount paid up is not less than 10 percent of the total amount paid up on all Shares that confer that right.

#### 5.6 When poll may be demanded

A poll may be demanded either before or after the vote is taken on a resolution.

### 5.7 Counting of votes

If a poll is taken:

- votes must be counted according to the votes attached to the Shares of each Shareholder present in person or by proxy or Representative and voting;
- the scrutineers shall be the auditor unless the auditor is unable or unwilling to act or unless the chairperson directs to the contrary in which case the scrutineers shall be appointed by the chairperson;
- c. the chairperson of the meeting shall finally determine in good faith the admission or rejection of any vote;
- d. the chairperson may declare the result of a poll when its outcome is known regardless of whether all votes have been counted;
- e. the chairperson may declare the result of the poll at or after the meeting.

1904396 4422040

### 5.8 Equality of votes

In the case of an equality of votes, whether voting is by voice or show of hands or poll, the chairperson of the meeting is entitled to a second or casting vote.

### 5.9 Proxy holder may demand poll

For the purposes of this paragraph 5, the instrument appointing a proxy to vote at a meeting of a Company confers authority to demand or join in demanding a poll and a demand by a person as proxy for a Shareholder has the same effect as a demand by the Shareholder.

### 5.10 Voting entitlement

Subject to any rights or restrictions for the time being attached to any Class of Shares, every Shareholder present in person or by proxy and voting by voice or on a show of hands has one vote.

### 5.11 Chairperson may demand poll

The chairperson may demand a poll on a resolution, either before or after a vote on such resolution, by voice or on show of hands.

#### 5.12 Withdrawal of demand

The demand for a poll may be withdrawn.

### 5.13 Poll to be taken as chairperson directs

Except as provided in paragraph 5.14 of this Schedule, if a poll is duly demanded it must be taken in such manner as the chairperson directs, and the result of the poll will be deemed to be the resolution of the meeting at which the poll was demanded.

### 5.14 Poll on election of chairperson

A poll demanded on the election of a chairperson or on a question of adjournment must be taken forthwith. A poll demanded on any other question may be taken at such time and place as the chairperson of the meeting directs, and any business other than that upon which a poll has been demanded may be proceeded with pending the taking of the poll.

### 6. PROXIES

### 6.1 Right to vote by proxy

A Shareholder may exercise the right to vote either by being present in person or by proxy.

## 6.2 Right of proxy to attend

A proxy for a Shareholder is entitled to attend and be heard at a meeting of Shareholders as if the proxy were the Shareholder.

1904396 4422040

### 6.3 Appointment of proxy

A proxy must be appointed by notice in writing signed by the Shareholder and the notice must state whether the appointment is for a particular meeting or a specified term not exceeding 12 months.

### 6.4 Proxy form to be sent with Notice of Meeting

A proxy form must be sent to all <u>Shareholders of the Company with each notice</u> calling a meeting of the Company.

### 6.5 Two Way Voting

The proxy form shall as a minimum, (so far as the subject-matter and form of the resolution reasonably permits) provide for two-way voting (for or against) on all resolutions, enabling the <u>Share</u>holder to instruct the proxy as to the casting of the vote and may provide for the Shareholder to abstain from voting on each resolution and/or for the Proxy to exercise a discretion to vote (for or against) each resolution and should state clearly the consequences if no Proxy instruction is provided.

#### 6.6 Proxy form must not name proxy

The Company shall not issue any form of notice of appointment with a proxy named in it, either by name or by reference to an office which that proxy holds, but the Company may indicate in a footnote that certain persons or officers are willing to act as a proxy if a <u>Share</u>holder desires to appoint them or any of them.

#### 6.7 Validity of vote

A vote given in accordance with the terms of an instrument of proxy is valid notwithstanding the previous death or insanity of the Appointer or revocation of the proxy or revocation of the authority under which the proxy was executed, or the Transfer of any Share in respect of which the proxy is given, if no notice in writing of such death, insanity, revocation or Transfer has been received by the Company before the start of the meeting or adjourned meeting at which the proxy is used.

### 6.8 Deposit of Proxy

No proxy is effective in relation to a meeting unless the proxy notice is received at the place nominated in the notice of meeting not later than 48 hours before the start of the meeting.

### 7. POSTAL VOTES

7.1 Shareholders may not exercise the right to vote at a meeting by casting a postal vote unless the Board elects to the contrary in relation to a particular meeting. <u>A postal vote may be cast using electronic means permitted by the Board.</u>

### 8. MINUTES

#### 8.1 Minutes must be kept

The Board must ensure that minutes are kept of all proceedings at meetings of Shareholders.

1904396 4422040

Del	eted: quoted Security	
Del	eted:	
Del	eted: Security	
	,	

Deleted: Security

#### 8.2 Evidence

Minutes which have been signed correct by the chairperson of the meeting are prima facie evidence of the proceedings.

### 9. SHAREHOLDER PROPOSALS

#### 9.1 Notice of matter for discussion or resolution

A Shareholder may give written notice to the Board of a matter the Shareholder proposes to raise for discussion or resolution at the next meeting of Shareholders at which the Shareholder is entitled to vote.

### 9.2 Notice of Shareholder proposal at Company's expense

If the notice is received by the Board not less than 20 <u>Business</u> Days before the last day on which notice of the relevant meeting of Shareholders is required to be given by the Board, the Board must, at the expense of the Company, give notice of the Shareholder proposal and the text of any proposed resolution to all Shareholders entitled to receive notice of the meeting.

#### 9.3 Notice of Shareholder proposal at Shareholder's expenses

If the notice is received by the Board not less than 5 <u>Business</u> Days and not more than 20 <u>Business</u> Days before the last day on which notice of the relevant meeting of Shareholders is required to be given by the Board, the Board must, at the expense of the Shareholder, give notice of the Shareholder proposal and the text of any proposed resolution to all Shareholders entitled to receive notice of the meeting.

### 9.4 Notice of late Shareholder proposal to be given if practicable

If the notice is received by the Board less than 5 <u>Business</u> Days before the last day on which notice of the relevant meeting of Shareholders is required to be given by the Board, the Board may, if practicable, and at the expense of the Shareholder, give notice of the Shareholder proposal and the text of any proposed resolution to all Shareholders entitled to receive notice of the meeting.

### 9.5 Proposing Shareholder's written statement

If the Directors intend that Shareholders may vote on the proposal by proxy or by postal vote, they must give the proposing Shareholder the right to include in or with the notice given by the Board a statement of not more than 1000 words prepared by the proposing Shareholder in support of the proposal, together with the name and address of the proposing Shareholder.

### 9.6 Limits on obligation to include statement

The Board is not required to include in or with the notice given by the Board a statement prepared by a Shareholder which the Directors consider to be defamatory, frivolous, or vexatious.

#### 9.7 Payment by Shareholder of costs

Where the costs of giving notice of the Shareholder proposal and the text of any proposed resolution are required to be met by the proposing Shareholder, the proposing

1904396 4422040

Deleted: Working Deleted: Working

Deleted: Working

Deleted: Working

Shareholder must, on giving notice to the Board, deposit with the Company or tender to the Company a sum sufficient to meet those costs.

### 10. CORPORATIONS MAY ACT BY REPRESENTATIVES

A body corporate which is a Shareholder may appoint a Representative to attend a meeting of Shareholders on its behalf in the same manner as that in which it could appoint a proxy provided that the chairperson of a meeting, the Board, or the persons checking the entitlement of people to attend a meeting, shall waive any time limit for prior notice in respect of a corporation in favour of a person who at a meeting can produce reasonable evidence of their authority to represent the corporation. Until notice of revocation of such authority shall have been given to the Company any person so authorised shall be entitled to exercise the same powers on behalf of the corporation which such person represents as that corporation could exercise if it were an individual member of the Company.

#### 11. VOTES OF JOINT HOLDERS

Where 2 or more persons are registered as the holder of a Share, the vote of the person named first in the Share Register and voting on a matter must be accepted to the exclusion of the votes of the other joint holders.

### 12. LOSS OF VOTING RIGHTS

### 12.1 No vote when amount owing on Share

A Shareholder is not entitled to vote at any meeting of Shareholders (including a meeting of an Interest Group) in respect of any Share if any amount is due and payable on that Share by the Shareholder to the Company.

### 12.2 Voting restrictions

Notwithstanding anything to the contrary in this Constitution or the Listing Rules, a person is not entitled to cast a vote on a resolution when that person is disqualified from voting by virtue of the voting restrictions specified in the Listing Rules.

### 12.3 Deadline for challenge

Without prejudice to any remedy (other than those which take legal effect against the Company) which any <u>Share</u>holder may have against any disqualified person who casts a vote at a meeting in breach of paragraph 12.2 of this Schedule, no resolution of, or proceeding at, that meeting may be impugned on the basis of a breach of that paragraph. Any objection by a <u>Share</u>holder to the accuracy or completeness of any list of <u>Share</u>holders who are disqualified from voting on a resolution pursuant to paragraph 12.2 of this Schedule, which has been supplied by the Company to the <u>NZX</u> or to any <u>Share</u>holder, on request pursuant to the Listing Rules shall be disregarded by the Company and the chairperson of the relevant meeting if it is notified to the Company later than one Business Day before the time fixed for commencement of the meeting.

Deleted: of	Securities		]
Deleted: of	Securities		٦
Deleted: of	Securities		]
Deleted: Exc	change		]
Deleted: of	Equity Securities	3	]

Deleted: full

1904396 4422040

- 39 -

## SECOND SCHEDULE

### PROCEEDINGS OF THE BOARD

## 1. CHAIRPERSON

### 1.1 Election of chairperson

The Directors may elect one of their number as chairperson of the Board.

### 1.2 Terms of office

The Director elected as chairperson holds that office until he or she dies or resigns or the Directors elect a chairperson in his or her place.

#### 1.3 Election of chairperson for particular meetings

If no chairperson is elected, or if at a meeting of the Board the chairperson is not present within 5 minutes after the time appointed for the commencement of the meeting, the Directors present may choose one of their number to be chairperson of the meeting.

### 2. NOTICE OF MEETING

#### 2.1 Convening meetings

A Director or an Employee, if requested by a Director to do so, may convene a meeting of the Board by giving notice in accordance with paragraph 2.2 of this Schedule.

#### 2.2 Notice

The following provisions apply in relation to meetings of the Board (except where otherwise agreed by all Directors in relation to any particular meeting or meetings or as provided in paragraph 2.3 of this Schedule):

- a. not less than two clear days' notice of a meeting of the Board shall be sent to each Director and to an Alternate Director required to receive notice pursuant to clause <u>21.2</u>, unless the Director waives that right, or a shorter period of notice is required to enable the Board to comply with its obligations under the Listing Rules.
- b. notice to a Director of a meeting of the Board may be:
  - i. delivered to the Director;
  - ii. posted to the address given by the Director to the Company for such purpose; <u>or</u>
  - iv. sent by electronic means in accordance with any request made by the Director from time to time for such purpose.
- c. a notice of meeting shall specify the date, time and place of the meeting and, in the case of a meeting by means of audio, or audio and visual, <u>or electronic</u> communication, the manner in which each Director may participate in the proceedings of the meeting.

Deleted: that

 ${\bf Deleted:}$  iii. . sent by facsimile transmission to the facsimile telephone number given by the Director to the Company for such purpose; or .

1904396 4422040

- 41
- d. a notice given to a director pursuant to this paragraph 2.2 is deemed to be given:
  - i. in the case of delivery, by handing the notice to the Director or by delivery of the notice to the address of the Director;
  - ii. in the case of posting, 3 days after it is posted; or
  - iv. in the case of electronic means, at the time of transmission.

### 2.3 Irregularity in notice

An irregularity in the notice of the meeting is waived if all Directors <u>entitled to receive</u> <u>notice of the meeting</u> attend the meeting without protest as to the irregularity or if all Directors entitled to receive notice of the meeting agree to the waiver.

### 3. METHOD OF HOLDING MEETINGS

3.1 A meeting of the Board may be held either:

- a. by a number of Directors sufficient to form a quorum being assembled together at the place, date and time appointed for the meeting; or
- b. by means of audio, audio and visual, or electronic communication by which all the Directors participating in the meeting and constituting a quorum can simultaneously hear each other throughout the meeting.

### 4. QUORUM

### 4.1 Number constituting quorum

A quorum for a meeting of the Board is three Directors.

### 4.2 No business without quorum

No business may be transacted at a meeting of Directors if a quorum is not present.

#### 4.3 Alternate Director may be included

In accordance with clause <u>21.1</u> of this Constitution an Alternate Director entitled to be present at a meeting shall be included for the purpose of establishing a quorum.

### 4.4 Interested Director

A Director shall not be counted in the quorum for the purpose of consideration of a matter in which the Director is interested (as defined in section 139 of the Act) unless the matter is one in respect of which Directors are expressly required by the Act to sign a certificate.

### 5. VOTING

5.1 Number of votes

1904396 4422040

**Deleted:** iii... in the case of facsimile transmission, when the Company receives a transmission report by the sending machine which indicates that the facsimile was sent in its entirety to the facsimile telephone number given by the Director;

Deleted: 23.1

Deleted: telephone, Deleted: or Every Director has one vote.

#### 5.2 Chairperson has casting vote

Subject to any restriction in the Listing Rules, the chairperson has a casting vote.

### 5.3 Majority

A resolution of the Board is passed if it is agreed to by all Directors present without dissent or if a majority of the votes cast on it are in favour of it.

### 5.4 Presumption as to voting

A Director present at a meeting of the Board is presumed to have agreed to, and to have voted in favour of a resolution of the Board, unless he or she expressly dissents from, or votes against the resolution at the meeting.

### 5.5 Disclosure of Interests

A Director shall comply with the provisions of section 140 of the Act (relating to disclosure of interest of directors) but failure to comply with that section does not affect the operation of paragraph 5.6 of this Schedule.

### 5.6 Personal involvement of Directors

Notwithstanding any rule of law or equity to the contrary, but subject to sections 107(3) and 141 of the Act and section 199(2) of the Act, a Director may:

- a. contract with the Company in any capacity;
- b. be a party to any transaction with the Company;
- have any direct or indirect personal involvement or Interest in any transaction or arrangement to which the Company is a party or in which it is otherwise directly or indirectly interested or involved;
- d. become a director or other officer of, or otherwise be Interested in, any corporation promoted by the Company or in which the Company may be directly or indirectly interested as a shareholder or otherwise; and
- e. retain any remuneration, profit or benefits in relation to any of the foregoing;

and no contract or arrangement of any kind referred to in this clause may be avoided by reason of a Director's Interest.

#### 5.7 Interested Directors may not vote

A Director who is Interested in a transaction entered into, or to be entered into, by the Company:

a. may attend a meeting of the Board at which any matter relating to the transaction arises but shall not be included among the Directors present at the meeting for the purposes of a quorum and may not vote on a Board Resolution in respect of any matter relating to the transaction except as provided in paragraph 5.8 of this Schedule;

1904396 4422040

	- 43 -	
	b. may sign a document relating to the transaction on behalf of the Companiand may do any other thing in his or her capacity as a Director in relation the transaction, as if the Director were not interested in the transaction.	
5.8	Exception to voting prohibition	
	Notwithstanding the provisions of paragraph 5.7a of this Schedule, a Director may vo in respect of a matter in which he or she is Interested if that matter is one in respect which, pursuant to an express provision of the Act, Directors are required to sign certificate or one which relates to the grant of an indemnity pursuant to section 162 the Act.	of a
5.9	Alternate Director may vote	
	An Alternate Director may attend and vote at meetings of the Board in accordance wi and subject to clause <u>21.2</u> of this Constitution if the Director that has appointed the Alternate Director is absent from the meeting.	
6.	MINUTES	
	The Board must ensure minutes are kept of all proceedings at meetings of the Board.	Deleted: ensure minutes
7.	RESOLUTIONS	
7.1	Written resolution	
	A resolution in writing, signed or assented to by <u>all</u> Directors then entitled to recein notice of a Board meeting, is as valid and effective as if it had been passed at a meetin of the Board duly convened and held.	
7.2	Forms of resolution	
	Any such resolution may consist of several documents (including those transmitted electronic means) in like form each signed or assented to by one or more Directors.	Deleted: facsimile or other similar means of communication
7.3	Resolution to be kept in minute book	
	A copy of any such resolution must be entered in the minute book of Boa proceedings.	rd
8.	NO NOTICE TO DIRECTORS OUTSIDE NEW ZEALAND	
	It is not necessary to give notice of a meeting of the Board to any Director for the tin being absent from New Zealand but if a Director is resident outside New Zealand, or the knowledge of the Company is temporarily absent from New Zealand and the Director has appointed an Alternate Director under the provisions of this Constitution notice must (subject to clause <u>21.2</u> of this Constitution) be given to the Alternat Director.	to ne n,
9.	COMMITTEES	
	The proceedings of committees of the Board shall be governed by this Schedule with necessary modifications, provided that the quorum for such meetings shall be a major of the members of the committee or where there are more than 3 members of a su committee, the quorum shall be 2 members of the committee present or participating the meeting.	ty b-

1904396 | 4422040

# 10. OTHER PROCEEDINGS

Except as provided in this Schedule the Board may regulate its own procedure.

1904396 | 4422040

Page 1: [1] Deleted

AL

11/06/2019 16:03:00

Certified as the Constitution of the Company

Chairman of the Company

Page 6: [2] I	Page 6: [2] Deleted   AL		0
PART A:	INTRODUCTION		1
1. 1.1 1.2	DEFINITIONS AND INTERPRETATION DEFINED TERMS INTERPRETATION		<b>1</b> 1 6
2. 2.1 2.2 2.3 2.4	APPLICATION OF THE EXCHANGE LISTING RULES COMPLIANCE WITH LISTING RULES EFFECT OF RULINGS BY THE EXCHANGE EFFECT OF FAILURE TO COMPLY WITH LISTING RULES OBLIGATIONS PRIOR TO LISTING		<b>7</b> 7 7 8
PART B:	SHARES AND SHAREHOLDERS		8
3. 3.1 3.2 3.3	<b>RIGHTS ATTACHING TO SHARES</b> CLASSES OF SHARES REDEEMABLE SHARES TERMS OF ISSUE OF SERIES A CONVERTIBLE PREFERENCE SHARES		<b>8</b> 8 8
$\begin{array}{c} 4.\\ 4.1\\ 4.2\\ 4.3\\ 4.4\\ 4.5\\ 4.6\\ 4.7\\ 4.8\\ 4.9\\ 4.10\\ 4.11\\ 4.12\\ 4.13\\ 4.14\\ 4.15\\ 4.16\\ 4.17\\ 4.18\\ 4.19\end{array}$	ISSUE OF SHARES AND OTHER SECURITIES BOARD HAS POWER TO ISSUE SHARES AND OTHER SECURITIES PRE-EMPTIVE RIGHTS ON ISSUE PROHIBITION ON ISSUE SPECIFICALLY APPROVED ISSUES PRO RATA AND \$5,000 OFFERS ISSUES WITHIN 15% LIMIT EMPLOYEE SHARE ISSUES OTHER PERMITTED ISSUES OTHER PERMITTED ISSUES TREASURY STOCK TRANSACTIONS DEEMED TO BE ISSUES OF EQUITY SECURITIES ISSUE OF SECURITIES AFFECTING CONTROL TRANSFER OF RIGHTS CONSOLIDATION AND SUBDIVISION OF SHARES ENTITLEMENTS TO THIRD PARTY SECURITIES NOTICE OF MEETINGS INTENTIONALLY BLANK PARTLY PAID SECURITIES PARTICIPATION OF OPTIONS IN RIGHTS ISSUES OPTION - CHANGE OF EXERCISE PRICE OR NUMBER OF UNDERLYING	1 1 1 1 1 1 1 1 1 1 1 1 1 1	<b>9</b> 9 9 9 10 11 11 12 13 13 14 14 14 15 15

4.20 4.21	CHANGE OF OPTION ON A BONUS ISSUE CHANGE OF OPTION ON CONSOLIDATION OR SUBDIVISION	15 15
5.	MODIFICATION OF SHAREHOLDER RIGHTS	16
5.1	PROCEDURE FOR MODIFICATION OF SHAREHOLDER RIGHTS	16
5.2	EXCEPTIONS	16
5.3	ISSUE OF EQUAL RANKING EQUITY SECURITIES	16
5.4	CANCELLATION OF UNPAID AMOUNTS	16
5.5	ACTIONS NOT INVALID	16
6.	CALLS ON SHARES	17
6.1	BOARD MAY MAKE CALLS	17
6.2		17
6.3	FIXED INSTALMENT DEEMED CALLS	17
6.4		17
6.5 6.6	DIFFERENT AMOUNTS	17
6.6 6.7	LIABILITY INTEREST AND EXPENSES	17 17
6.8	PROCEEDINGS FOR RECOVERY OF CALL	18
6.9	PAYMENT IN ADVANCE OF CALLS	18
7.	LIEN ON SHARES	18
7.1	LIEN ON SHARES	18
7.2	WAIVER OF LIEN	18
8.	FORFEITURE OF SHARES	18
8.1	NOTICE REQUIRING PAYMENT OF CALL	18
8.2	CONTENTS OF NOTICE	19
8.3	FORFEITURE FOR NON-PAYMENT	19
8.4		19
8.5	CANCELLATION OF FORFEITURE	19
8.6	EFFECT OF FORFEITURE	19
9.	SALE OF SHARES SUBJECT TO LIEN OR FORFEITURE	19
9.1	POWER OF SALE	19
9.2	APPLICATION OF PROCEEDS OF SALE	20
9.3 9.4	EVIDENCE THAT POWER OF SALE HAS ARISEN SALE PROCEDURE	20 20
10.	DISTRIBUTIONS TO SHAREHOLDERS	20
10.1	SATISFACTION OF SOLVENCY TEST	20
10.2	RESTRICTION ON CERTAIN DISTRIBUTIONS	20
10.3	METHOD OF PAYMENT	21
10.4	SHARES IN LIEU OF DIVIDENDS	21
10.5	BONUS ISSUES	21
10.6	SHAREHOLDER DISCOUNTS	22
10.7		22
10.8	NO INTEREST	22 22
10.9	UNCLAIMED DISTRIBUTIONS	22
11.	ACQUISITION AND REDEMPTION OF COMPANY'S OWN SHARES	<b>23</b>
11.1	COMPANY MAY PURCHASE, ACQUIRE OR REDEEM ITS OWN SHARES	23

11.2 11.3	PROHIBITIONS ON PURCHASE OR ACQUISITION PRIOR NOTICE OF ACQUISITION	23 24
11.4	Acquisition or Redemption with Approval of Equity Security Holders	24
11.5	Acquisition of Equity Securities other than Shares	25
12.	FINANCIAL ASSISTANCE BY THE COMPANY TO PURCHASE ITS SHARE	ES 25
12.1	FINANCIAL ASSISTANCE	25
12.2	PERMITTED FINANCIAL ASSISTANCE	25
12.3	FINANCIAL ASSISTANCE WITH APPROVAL OF EQUITY SECURITY HOLDERS	26
13.	TRANSFER OF SHARES	26
13.1	TRANSFERABILITY	26
13.2		26
13.3	TRANSFERS EXECUTED OVERSEAS	26
13.4	OTHER FORMS OF TRANSFER DELIVERY TO COMPANY	26
13.5 13.6	POWER TO REFUSE OR DELAY REGISTRATION	27 27
13.7	COMPULSORY DISPOSAL WHEN HOLDING LESS THAN MINIMUM HOLDING	27
13.8	When Transfer Effective	28
13.9	MULTIPLE REGISTERS	28
13.10	TRANSFER OF RIGHTS	28
14.	TRANSMISSIONS AND EQUITABLE INTERESTS	28
14.1	TRANSMISSION ON DEATH OF SHAREHOLDER	28
14.2	RIGHTS OF PERSONAL REPRESENTATIVES	28
14.3	JOINT PERSONAL REPRESENTATIVES	28
14.4	NO NOTICE OF TRUSTS	29
14.5	NO RECOGNITION OF EQUITABLE INTERESTS	29
15.	SEPARATE PARCELS OF SHARES	29
15.1	SEPARATE PARCELS	29
16.	SHAREHOLDERS REPORTS, INFORMATION AND NOTICES	29
16.1	COMPANY REPORTS AND FINANCIAL STATEMENTS	29
16.2	INSPECTION OF COMPANY RECORDS	29
16.3	INSPECTION BY SHAREHOLDERS	30
16.4 16.5	METHOD OF SERVICE OF NOTICES, REPORTS AND OTHER DOCUMENTS SERVICE OF NOTICES OVERSEAS	30 30
16.6	ACCIDENTAL OMISSIONS	30
16.7	JOINT SHAREHOLDERS	30
17.	MEETINGS OF SHAREHOLDERS	30
17.1	ANNUAL MEETINGS	30
17.2	TIME AND PLACE OF ANNUAL MEETING	31
17.3	SPECIAL MEETINGS	31
17.4	CALLING OF SPECIAL MEETINGS	31
17.5	EQUITY SECURITY HOLDERS ENTITLED TO ATTEND	31
17.6	METHODS OF HOLDING MEETINGS	31
18.	PROCEDURE AT SHAREHOLDER MEETINGS	31

19. 19.1 19.2 19.3 19.4	RESTRICTION ON DISPOSAL OR ACQUISITION OF ASSETS RESTRICTION CONDITIONAL TRANSACTION EXCEPTION NOTICE OF MEETING RELATING TO DISPOSAL OR ACQUISITION OF ASSETS	<b>31</b> 32 32 32
20. 20.1 20.2	<b>RESTRICTION ON TRANSACTIONS WITH RELATED PARTIES</b> RESTRICTION CONDITIONAL AGREEMENT INVOLVING A MATERIAL TRANSACTION	<b>33</b> 33 33
20.3 20.4	PERMITTED EXCEPTIONS NOTICE OF MEETING RELATING TO MATERIAL TRANSACTION WITH A RELATED PARTY	33 Y34
PART C:	DIRECTORS	34
21.	POWERS OF DIRECTORS	34
21.1	MANAGEMENT OF COMPANY	34
21.2	EXERCISE OF POWERS BY BOARD	34
21.3	DELEGATION OF POWERS	35
21.4	APPOINTMENT OF ATTORNEY	35
21.5	RATIFICATION BY SHAREHOLDERS	35
22.	APPOINTMENT AND REMOVAL OF DIRECTORS	35
22.1	NUMBER OF DIRECTORS AND RESIDENCE	35
22.2	APPOINTMENT	35
22.3	EXISTING DIRECTORS TO CONTINUE	35
22.4	NOMINATION OF DIRECTORS	36
22.5	RESTRICTION ON APPOINTMENT OF SEVERAL DIRECTORS BY SINGLE RESOLUTION	36
22.6	ROTATION OF DIRECTORS	36
22.7	EXCEPTIONS TO ROTATION	36
22.8	RE-ELECTION OF RETIRING DIRECTOR	36
22.9	VACATION OF OFFICE	37
22.10	APPOINTMENT BY BOARD	37
22		27
23.		37
23.1	APPOINTMENT	37
23.2	RIGHTS OF ALTERNATE DIRECTOR	37
23.3	REMUNERATION AND EXPENSES	38
23.4	CESSATION OF APPOINTMENT	38
24.	EXECUTIVE DIRECTOR	38
24.1	APPOINTMENT	38
24.2	Powers	39
24.3	NOT SUBJECT TO RETIREMENT BY ROTATION	39
24.4	RESIGNATION	39
24.5	DISMISSAL	39
24.6	REMUNERATION	39
25.	DIRECTORS REMUNERATION AND OTHER BENEFITS	39
25.1	Authorisation	39
25.2	INTERESTS REGISTER AND CERTIFICATE	40
25.3	SHAREHOLDER APPROVAL	40
25.4	POWER TO INCREASE WHEN ADDITIONAL DIRECTOR	40
20.4		40

25.5 25.6	NOTICE OF PROPC		41 41		
26. 26.1 26.2 26.3 26.4 26.5	INDEMNITY AND INDEMNITY OF DIR OTHER INDEMNITH EXCEPTIONS INSURANCE DEFINITIONS	ECTORS	<b>41</b> 41 42 42 42		
27.	PROCEEDINGS	OF THE BOARD	42		
PART D	: GENERAL		43		
28. 28.1 28.2 28.3	<b>METHOD OF CO</b> Deeds Other written C Other obligatio	CONTRACTS	<b>43</b> 43 43 43		
29. 29.1 29.2	ACCOUNTS ACCOUNTS TO BE APPOINTMENT OF		<b>43</b> 43 43		
30. 30.1 30.2	LIQUIDATION DISTRIBUTION OF A PREFERENCE ON I	ASSETS LIQUIDATION TO SERIES A CONVERTIBLE PREFERENCE SHARES	<b>43</b> 43 44		
31.	AUDIT COMMIT	ſEE	44		
FIRST S	CHEDULE				
	PROCEEDINGS	AT MEETINGS OF SHAREHOLDERS	45		
SECON	SECOND SCHEDULE				
	PROCEEDINGS	OF THE BOARD	53		
Page 1: [3]	Deleted	AL 11/06/2019 18:49	9:00		
	Aggregate Net Value"	in relation to assets has the meaning given to the term in Listing Rule 9.2.2.			

	0	
Page 1: [4] Deleted	AL	11/06/2019 18:49:00
"Appraisal Report" has the meaning given to that term by Listing Rule 1.2.		g Rule 1.2.
"Associated Person" has the meaning given to that term by Listing R		g Rule 1.3.
"Audit Committee"	has the meaning given to that term by Listing	g Rule 1.1.2.

"Average Market	has the meaning given to that term by Listing Rule 1.1.2.	
Page 1: [5] Deleted	AL	11/06/2019 18:50:00
"Disqualifying	has the meaning set out in Listing F <b>Relationship</b> "	Rule 1.1.2.
Page 2: [6] Deleted	AL	11/06/2019 19:00:00
"Material Transaction"	means, for the purposes of clause a related series of transactions in t or any of its subsidiaries:	
a.	purchases or otherwise acquires, lessee) or sells or otherwise dis Aggregate Net Value in excess o Capitalisation of the Company; or	poses of, assets having an
b.	issues its own securities or acquir having a market value in excess o Capitalisation of the Company; or	
С.	borrows, lends, pays, or receiv obligation, of an amount in exc Market Capitalisation of the Comp	ess of 5% of the Average
d.	enters into any guarantee, indem obligation, or gives any security, could expose the Company and excess of 5% of the Average I Company; or	for or of obligations which its subsidiaries to liability in
e.	provides or obtains any services (including, without limitation, obtaining underwriting of Securities or services as an employee) in respect of which the actual gross cost to the Company and its subsidiaries in any financial year (ignoring any returns or benefits in connection with such services) is likely to exceed an amount equal to 0.5% of the Average Market Capitalisation of the Company; or	
f. Subsidiary of the Compan the Company.	amalgamates, except for amalga y with another wholly owned Subsidia	-
Page 2: [7] Deleted	AL	13/06/2019 15:03:00
"month"	means a calendar month.	
Page 2: [8] Deleted	AL	11/06/2019 19:09:00
"Ordinary of the Company which carry	means a resolution that is approved <b>Resolution</b> " of the Votes of Votes entitled to Vote and voting on th	of those holders of Securities
Page 3: [9] Deleted	AL	11/06/2019 18:57:00
"Ruling"	means any decision or determina	

**"Ruling"** means any decision or determination by the Exchange as to the meaning or interpretation or application of the Rules and includes any filing, waiver or revocation of a waiver given pursuant to Listing Rule 1.4, 1.6 or 1.7.

Page 3: [10] Deleted	AL	11/06/2019 19:12:00
----------------------	----	---------------------

"Security"	means any interest or right to particip earnings, royalties, or other prope includes:	
а.	any renewal or variation of the ter existing security;	rms or conditions of any
	b. any Debt Security; and	
	c. any Option or Right.	
Page 3: [11] Deleted	AL	11/06/2019 19:14:00
"Share Register"	means the share register for the Cor in accordance with the Act.	mpany required to be kept
Page 3: [12] Deleted	AL	11/06/2019 18:50:00
"Subsidiary"	has the meaning given to the term in	Listing Rule 1.1.2.
"Trading Participant"	has the meaning given to that term by	y Listing Rule 1.1.2.
"Treasury Stock"	has the meaning given to that term by	y Listing Rule 1.1.2.
"Vote"	has the meaning given to that term b	y Listing Rule 1.1.2.
Page 4: [13] Deleted	AL	11/06/2019 19:29:00
	defined in the Listing Rules shall, vides otherwise, have the same meani	
Page 5: [14] Deleted	AL	11/06/2019 20:12:00
Obligations prior to I	Listing	
For as long as the Co	ompany is not listed on the Exchange:	
a. any provision of this Constitution which states that it is subject to the approval of the Exchange shall be read as if that approval had been obtained; and		
<ul> <li>all references to Quoted Equity Securities shall be read as Equity Securities; and</li> </ul>		
c. Rulings of the Securities; and	Exchange are not binding on the Co	ompany or the holders of
d. any provision	or part of a provision that is not inter	nded to have effect if the
	ot so listed will not have effect.	

b. series A convertible preference Shares;

Page 5: [16] Deleted	AL	11/06/2019 20:25:00
	and the second state of th	and the state of t

subject to this Constitution and to any rights previously conferred on the holders of any existing Share or Class of Share.

Redeemable Shares

If redeemable Shares are issued then the terms on which those Shares are issued are deemed to be incorporated in this Constitution.

Page 5: [17] Deleted	AL	11/06/2019 20:17:00
Terms of Issue of Ser	ies A Convertible Preference Share	is

The series A convertible preference Shares are issued on the following terms:

- the Shares are convertible to ordinary Shares at the option of the holders on the following basis:
- that 180 days have passed from the date of issue of the Shares;
- the holder must give written notice to the Board to convert the Shares;
- the Board must convert these Shares to ordinary Shares within 10 days of receiving the written notice from a holder; and
- these Shares will be converted on the basis that one series A convertible preference Share converts to five ordinary Shares.
- the Shares will automatically convert into ordinary Shares on the basis that one series A convertible preference Share converts to five ordinary Shares when the Company issues further ordinary Shares at not less than \$0.75 per share that total \$25,000,000.00.
- the Shares are not transferable except to a Personal Representative of the holder and only on the basis that within 12 months of the Personal Representative being entitled to the Shares, they must convert them in accordance with clause 3.3.a except that clause 3.3.a.i will not apply.

Page 7: [18] Deleted	AL	13/06/2019 14:50:00
----------------------	----	---------------------

The Board may issue Equity Securities if:

- a. **Rights Issues**: those Equity Securities are offered to the holders of existing Equity Securities on a basis which, if the offer were accepted by all such holders, would maintain the existing proportionate rights of each existing holder (relative to other holders of Equity Securities) to Votes and to Distribution Rights, and that offer is Renounceable; or
- b. **Bonus Issues**: those Equity Securities are issued to the holders of existing Equity Securities as fully paid Securities on a basis which maintains the existing proportionate rights of each existing holder (relative to other holders of Equity Securities) to Votes and to Distribution Rights. However bonus issues may only be made to holders of ordinary shares if such an issue is also made to the holders of series A convertible preference Shares; or
- c. those Equity Securities are offered to all holders of existing Equity Securities of the Company carrying Votes for consideration not exceeding \$5,000 per existing Equity Security holder (being the registered holder or, in the case of Securities held through a custodian, the beneficial owners of the Securities) and the number of Equity Securities to be issued is not greater than 30% of

the number of fully paid Equity Securities carrying Votes that are already on issue;

provided that notwithstanding subclauses a., b. and c. of this clause 4.5:

Page 7: [19] DeletedAL	13/06/2019 14:50:00
------------------------	---------------------

- e. **Offer in Accordance with Specific Rights**: the Board may offer and issue Equity Securities to the holders of existing Securities in accordance with specific rights attached to those existing Securities to participate in issues of Equity Securities, notwithstanding that the effect may be that existing proportionate rights to Votes and Distribution Rights are not maintained; and
- f. **Rounding up to Minimum Holding**: the Board may authorise a disproportionate offer to the extent necessary to round up holdings of Equity Securities to a Minimum Holding, or to avoid the creation of holdings which are not Minimum Holdings; and
- g. **No Offer if Expressly Excluded**: the Board shall not offer or issue Equity Securities to holders of existing Equity Securities the terms of which expressly exclude the right to participate in the relevant offer or issue; and

h. **No Offer Outside New Zealand**: the Board shall not offer or issue Equity Securities to holders of existing Securities in a jurisdiction outside New Zealand if the legal requirements of that jurisdiction are such that it is unduly onerous for the Company to make the offer in that jurisdiction provided that in the case of Renounceable Rights, the Company shall arrange the sale of any Renounceable Rights to the relevant Equity Securities and to account to holders in that jurisdiction for the proceeds.

D	7.	1001	D.1.4. J
Page	7:	20	Deleted

AL

13/06/2019 14:51:00

# **Issues within 15% Limit**

The Board may issue Equity Securities if:

- a. **Excluded Entities**: the issue is not made in whole or in part to any Director of the Company, Associated Person of a Director or Employee of the Company; and
- b. **Limited Percentage**: the total number of Equity Securities issued, together with all other Equity Securities of the same Class issued pursuant to this clause 4.6 during the shorter of the period of 12 months preceding the date of the issue and the period from the date on which the Company was Listed on the Exchange to the date of the issue, will not exceed the aggregate of:
  - i. 15% of the total number of Equity Securities of that Class on issue at the commencement of that period; and
  - ii. 15% of the number of Equity Securities of that Class issued during that period pursuant to any of clauses 4.4, 4.5, 4.7 and 4.8; and
  - iii. any Equity Securities of that Class issued pursuant to this clause 4.6 during that period, the issue of which has been ratified by an Ordinary Resolution; and

less:

iv. 15% of the number of Equity Securities of that Class which have been acquired or redeemed by the Company during that period (other than Equity Securities held as Treasury Stock).

For the purposes of this clause 4.6:

- v. Securities which will, or may, Convert to other Equity Securities are deemed to be of the same Class as, and to correspond in number to, the Equity Securities into which they will, or may Convert; and
- vi. where the Conversion ratio is fixed by reference to the market price of the underlying Equity Securities, the market price shall, unless otherwise specified in the terms of the issue, be the volume weighted average market price over the 20 Business Days before the earlier of the day the issue is made or announced to the market.

	Page 7: [21] Deleted	AL	13/06/2019 14:51:00
--	----------------------	----	---------------------

# **Employee Share Issues**

The Board may issue Equity Securities if the issue is of a Class of Securities already on issue and is made to Employees of the Company and:

- a. the total number of Securities issued, together with all other Equity Securities of the same Class issued to Employees pursuant to this clause 4.7 during the shorter of the period of 12 months preceding the date of the issue and the period from the date on which the Company was Listed on the Exchange to the date of the issue does not exceed 3% of the aggregate of:
  - i. the total number of Equity Securities of that Class on issue at the commencement of that period; and
  - ii. the total number of Equity Securities of that Class issued during that period pursuant to any of clauses 4.4, 4.5, 4.6 and 4.8;
- b. no Director or Associated Person of a Director shall participate in any such issue unless the scheme for such participation and the precise level of entitlement for each such person have previously been approved by an Ordinary Resolution.

Page 7: [22] Deleted	AL	13/06/2019 14:51:00

For the purposes of this clause 4.7:

- c,. an issue to a Director, or an Associated Person of a Director, solely in that person's capacity as a trustee of a bona fide employee share scheme, superannuation scheme, or the like, in which that Director or Associated Person has no beneficial interest, is deemed not to be an issue to a Director or Associated Person of a Director, or an issue in which Directors or Associated Persons participate;
- d. Securities which will, or may, Convert to other Equity Securities are deemed to be of the same Class as, and to correspond in number to, the Equity Securities into which they will, or may, Convert and where the conversion ratio is fixed by reference to the market price of the underlying Equity Securities, the market price, unless otherwise specified in the terms of issue,

shall be the volume weighted average market price over the 20 Business Days before the earlier of day the issue is made or announced to the market.

Except as provided in the Listing Rules the Company shall not reprice or amend the terms of any Securities issued with Shareholder approval to or for the benefit of Employees or Directors under Listing Rule 7.3 in their capacity as such, without either the approval of the Exchange or a further Ordinary Resolution of the Shareholders resolving to approve the repricing or amendment.

Page 7: [23] Deleted	AL	13/06/2019 14:52:00
Other Dermitted Jeause		

# Other Permitted Issues

The Board may issue Equity Securities in any of the following circumstances:

- a. **Takeovers**: the issue is made as consideration in an offer made by the Company or any of its subsidiaries in accordance with:
  - i. any takeover code approved under the Takeovers Act 1993; or
  - ii. the provisions of the constitution or other governing documentation of the Company which comply with the requirements of section 4 of the Listing Rules where the Company is not a Code Company; or
  - iii. any takeover law regime of a jurisdiction other than New Zealand which provides for prior notice, publicity and disclosure which in the opinion of the Exchange is at least as useful to the recipients of the offer as the requirements of one or more of the provisions referred to in paragraphs i. or ii. of this subclause; and

that offer is made to all holders (other than the Company and its Related Companies) of Equity Securities (as defined in the Listing Rules) in any company or other entity Listed on the NZSX, or on a Recognised Stock Exchange, which is not a company or other entity that is an Associated Person of the Company or of any Director; or

Page 7: [24] Deleted	AL	13/06/2019 14:52:00
Conversion of Conver	tible Securities: the issue is made	de upon Conversion of any
	or any other Security from tim terms of issue of those Securities rities issued; or	

- c. **Issue to Round Up Holdings:** the issue is made to a holder of Equity Securities in order to bring that person's holding up to a Minimum Holding; or
- d. **Issue on Amalgamation:** the issue is made pursuant to Part XIII or Part XV of the Act; or

e. **Issue in Lieu of Dividends:** the issue is made pursuant to a plan for the issue of Securities in lieu of Dividends or as part of a dividend re-investment plan that entitles an existing Shareholder to subscribe for Securities by applying all or any specified part of any dividend declared by the Company and payable to that person, and which issue or dividend re-investment plan would, except to the extent that the plan excludes existing holders in a jurisdiction outside New Zealand if the legal requirements of that jurisdiction are such that it is unduly onerous for the Company to extend the plan to that jurisdiction, maintain the existing right of each existing holder relative to other holders of Equity

Securities to Votes and Distribution Rights if the offer were accepted by all such holders. However any issue to holders of ordinary Shares must also be made to the holders of series A convertible preference Shares.

Page 7: [25] Deleted AL	13/06/2019 14:52:00
-------------------------	---------------------

# **Treasury Stock**

The transfer by the Company of Treasury Stock shall be deemed to be an issue of new Equity Securities to which the provisions of clause 4 apply.

Page 7: [26] Deleted	AL	13/06/2019 14:52:00
Transactions Deam	ad to be leaves of Equity Coourities	

# Transactions Deemed to be Issues of Equity Securities

The grant or creation by the Company in favour of the holders of any Class of Equity Securities of any right or entitlement to Securities of a third party (whether or not that third party is an Issuer of Securities), shall be deemed to be an issue of new Equity Securities to which the provisions of clause 4.3 apply.

Page 7: [27] Deleted	AL	13/06/2019 14:53:00

# **Issue of Securities Affecting Control**

Notwithstanding the provisions of clause 4.3, no issue, acquisition or redemption of Securities shall be made by the Company if:

- a. there is a significant likelihood that the issue, acquisition or redemption will result in any person or group of Associated Persons materially increasing their ability to exercise, or direct the exercise (either then or at any future time) of, effective control of the Company; and
- that person or group of Associated Persons is entitled before the issue, acquisition or redemption to exercise, or direct the exercise of, not less than 1% of the total votes attaching to the Securities of the Company;

unless the precise terms and conditions of the issue have been approved by an Ordinary Resolution of the Company.

# **Transfer of Rights**

The terms of issue of Equity Securities pursuant to clause 4.5a are subject to clause 13.10 of this Constitution.

Page 7: [28] Deleted	AL	11/06/2019 20:40:00
However any such consolidation	n or subdivision made to the class o	of ordinary Shares must also
be made to the class of series A	convertible preference Shares on t	the same basis.

Page 7: [29] Deleted   AL	13/06/2019 14:53:00
---------------------------	---------------------

# **Entitlements to Third Party Securities**

Entitlements conferred by the holding of Equity Securities of the Company, to Securities of a third party (whether or not that third party is an issuer as defined in the Listing Rules), shall not be conferred or created other than in accordance with clause 4, as if such Securities comprised an issue of Equity Securities of the Company.

Page 7: [30] Deleted	AL	13/06/2019 14:54:00

Notice of Meetings

- The text of any resolution to be put to a meeting for the purposes of clauses a. 4.4, 4.6.b.iii, 4.7 and 4.11 shall be set out in the notice of the relevant meeting. That notice shall be approved by the Exchange and in accordance with Listing Rule 6.1, and shall contain the particulars specified in Listing Rule 6.2.1.
  - b. A notice of meeting to consider a resolution of the nature referred to in subclause a. shall be accompanied by an Appraisal Report if:
    - in the case of an issue, the issue is intended or is likely to result in more than 50% of the Securities to be issued being acquired by Directors or Associated Persons of Directors. No Appraisal Report is required to accompany any notice of meeting referred to in subclause a. if the issue is made pursuant to clause 4.7 and the amount in terms of the proposed issue will be determined according to criteria applying generally to all Employees eligible to participate in that issue.
    - in the case of an acquisition or redemption or the giving of financial assistance, it is intended or likely that more than 50% of the Securities to be acquired or redeemed will be Securities held by Directors or Associated Persons of Directors of the Issuer, or that more than 50% of the total financial assistance to be given will be given to such persons.

# **Intentionally Blank**

# **Partly Paid Securities**

Each Security which is not fully paid shall carry only a fraction of the Vote which would be exercisable if the Security were fully paid. The fraction must be equivalent to the proportion which the amount paid (not credited) is of the total amounts paid and payable (excluding amounts credited and amounts paid in advance of a call).

Page 7: [31] Deleted	AL	13/06/2019 14:56:00
Participation of Optio	ns in Rights Issues	

Participation of Options in Rights Issues

An Option issued after the Company has Listed on the Exchange must not confer the right to participate in a Rights issue unless the Option:

- is exercised before the Record Date for the Rights issue; or a.
- was issued under a pro rata offer made pursuant to clause 4.5 to the holders of b. Quoted Equity Securities; or
- was issued with the approval of holders of Quoted Equity Securities and the C. Option holder can participate in a new issue to the holders of the underlying Securities in accordance with the terms of such an Option.

Page 7: [32] Deleted	AL	13/06/2019 14:57:00
Option - Change of Exercise Price or Number of Underlying Securities		ying Securities

The Company must not issue an Option after the Company is Listed on the Exchange which confers the right to a change in the exercise price or number of underlying Securities, except if that Option:

- a. was issued with the approval of holders of Quoted Equity Securities, then the exercise price or number of underlying Securities may change in accordance with the formula or provision contained in the terms of the Option if there is a Rights issue to the holder of the underlying Securities; or
- b. was not issued with the approval of holders of Quoted Securities and there is a Rights issue to the holders of the underlying Securities, then the exercise price of an Option may be reduced according to the formula set out in Listing Rule 8.1.7.

Page 7: [33] Deleted	AL	13/06/2019 14:56:00

# Change of Option on a Bonus Issue

If there is a bonus issue to the holders of the underlying Securities, the number of Securities over which an Option is exercisable may be increased (or additional Securities may be reserved for issue on exercise of an Option) by the number of Securities which the holder of the Option would have received if that Option had been exercised before the Record Date for the issue, consolidation or subdivision.

Page 7: [34] Deleted	AL	13/06/2019 14:56:00
----------------------	----	---------------------

# Change of Option on Consolidation or Subdivision

If there is a consolidation or subdivision or similar proportionate reconstruction of the underlying Securities the number of Securities over which an Option is exercisable may be consolidated or subdivided in the same ration and the exercise price amended in inverse proportion to that ratio.

Page 12: [35] Deleted	AL	11/06/2019 20:58:00
-----------------------	----	---------------------

# **Evidence that Power of Sale has Arisen**

A certificate signed by a Director stating that the power of sale has arisen and is exercisable by the Company, or that a Share has been forfeited on the date stated in the certificate shall be conclusive evidence of those facts.

Page 15: [36] Deleted	AL	13/06/2019 14:37:00

in accordance with the provisions of the Act and this Constitution

Page 15: [37] Deleted	AL	13/06/2019 14:37:00
Prohibitions on Purch	ase or Acquisition	

Subject to clause 11.3, the Company shall not purchase, acquire or redeem any of its Equity Securities other than by way of:

- a. an acquisition effected by offers made by the Company through the order matching market of the Exchange or of a Recognised Stock Exchange, or
- b. an acquisition effected in compliance with section 60(1)(a) (read together with section 60(2)) of the Act; or

- c. an acquisition of the nature referred to in section 61(7) of the Act; or
- d. an acquisition or redemption approved in accordance with clause 11.5; or
- e. an acquisition required by a Shareholder pursuant to section 110 or section 118 of the Act; or
- f. an acquisition effected in compliance with section 60(1)(b)(ii) (read together with section 61) of the Act and:
  - i. is not made from any person who is a Director, Associated Person of a Director, or Employee; and
  - ii. the total number of Equity Securities acquired, together with all other Equity Securities of the same Class as those Equity Securities that are to be acquired pursuant to this subclause f. during the shorter of the period of 12 months preceding the date of the acquisition and the period from the date in which the Company was Listed on the Exchange to the date of the acquisition, will not exceed 15% of the total number of Equity Securities of that Class on issue at the commencement of that period; or:

provided that for the purposes of this subclause f.:

- iii. Securities which will, or may, Convert to other Equity Securities shall be deemed to be of the same Class as, and to correspond in number to, the Securities into which they will, or may, Convert; and
- iv. Where the Conversion ratio of any such Securities is fixed by reference to the market price of the underlying Securities, the market price for the purposes of this subclause f shall be the volume weighted average market price over the 20 Business Days before the earlier of the day the acquisition is entered into and or announced to the market.
- g. a redemption from a holder who holds less than a Minimum Holding; or
- h. a redemption of Equity Securities issued in compliance with clause 4.4 or clause 4.5 where the Issuer is bound or entitled to redeem those Equity Securities pursuant to their terms of issue, or
- i. a redemption in compliance with section 69(1)(a) of the Companies Act 1993; or
- j. a redemption of Equity Securities that are Debt Securities which may be Converted into shares in an Issuer which is a company, and, before that Conversion, they are redeemed in cash.

# Prior Notice of Acquisition

Before the Company purchases or acquires any Equity Securities, other than a purchase or an acquisition from a holder who holds less than a Minimum Holding, the Company shall give at least three Business Days' notice to the Exchange specifying:

- a a period of time, not exceeding 12 months from the date of the notice, within which the Company will acquire the Equity Securities; and
- b. the Class and maximum number of Equity Securities to be acquired in that period;

provided that the Company may at any time cancel any notice so given, or, by three Business Days' notice to the Exchange, vary any such notice.

Page 15: [38] Deleted	AL	13/06/2019 14:39:00
Acquisition or Redem	otion with Approval of Equity S	ecurity Holders

- The Company may acquire or redeem Equity Securities under clause 11.2.d if the precise terms and conditions of the specific proposal (the "Proposal") to acquire or redeem those Equity Securities have been approved by separate resolutions (passed by a simple majority of Votes) of members of each separate group of each Class of Quoted Equity Securities whose rights or entitlements are materially affected in a similar way by that proposal; and a Proposal authorised by resolutions passed pursuant to clause 11.4(a) shall be completed:
- if that Proposal is transacted solely with Employees (as defined in Listing Rule 7.3.6) within 36 months after the passing of those resolutions; or
- in all other circumstances, within twelve months after the passing of those resolutions.

# Acquisition of Equity Securities other than Shares

Equity Securities which are not Shares may be acquired pursuant to clauses 11.2.b, 11.2.c and 11.2.f, or redeemed pursuant to clause 11.4.b, if the Company complies with the sections of the Act referred to in the relevant clause, on the basis that references in those sections of the Act to:

"shares" shall be deemed to be references to all Equity Securities of the Class of Equity Securities which is the subject of the acquisition or redemption and references to "shareholders" shall be read accordingly; and

"constitution" shall be deemed to be references to the document which governs the rights attaching to those Equity Securities.

Page 18: [39] Deleted	AL	13/06/2019 15:58:00

# Transfer of Rights

Every person to whom unissued Equity Securities are offered pursuant to clause 4.5a may decline or accept the offer, or Transfer their Rights thereunder to any person or persons to whom the Equity Securities, when issued, could be transferred but the Directors have the same right to decline to accept any such Transfer as they would have if the Transfer were a Transfer of Shares, and the provisions of this Constitution as to the Transfer of Shares, with all necessary modifications, apply to Transfers of Rights to unissued Equity Securities.

13/06/2019 14:10:00

Inspection of Company Records

The Company will keep the following Records available for inspection and copying in accordance with the Act by any Shareholder or by a person authorised in writing by a Shareholder who gives notice in accordance with the Act of intention to inspect:

- a. the certificate of incorporation of the Company;
- b. the Constitution;
- c. the Share Register;
- d. the names and residential addresses of the Directors;
- e. the registered office and address for service of the Company;
- f. minutes of all meetings and resolutions of Shareholders;
- g. copies of all written communications to all Shareholders or Class of Shareholders during the preceding 10 years, including annual reports, financial statements, and group financial statements;
- h. certificates given by the Directors under the Act; and
- i. the interests register of the Company.

Page 21: [41] Deleted	AL	13/06/2019 14:22:00
Postriation		

## Restriction

The Company shall not enter, or permit any of its subsidiaries to enter, into any transaction or series of linked or related transactions to acquire, sell, lease, exchange, or otherwise dispose of (otherwise than by way of charge) assets held or to be held by the Company:

- a. which would change the essential nature of the business of the Company and its subsidiaries taken as a whole; or
- b. in respect of which the gross value is in excess of 50% of the Average Market Capitalisation of the Company;

except with the prior approval of an Ordinary Resolution or a Special Resolution if the Company must obtain approval of the transaction or transactions under section 129 of the Act.

# Conditional transaction

The Company or a subsidiary may enter into an agreement involving a transaction which in terms of clause 19.1 requires the approval of an Ordinary Resolution or a Special Resolution if the agreement provides that the transaction is conditional upon the passing of such resolution and may not take place until the resolution has been passed.

Page 21: [42] Deleted	AL	13/06/2019 14:22:00
Exception		

# Clause 19.1 shall not apply to:

A takeover offer made by a Company:

- In respect of a Code Company in accordance with any takeovers code approved under the Takeovers Act 1993; or
- In respect of a company that is not a Code Company but to whom section 4 of the Listing Rules applies, in accordance with the Constitution or Trust Deed of that other company which complies with section 4 of the Listing Rules where that other company is not a Code Company; or
- In relation to any other person, in accordance with any takeover law regime of a jurisdiction other than New Zealand which is applicable to that person and provides for prior notice, publicity and disclosure which in the opinion of the Exchange is at least as useful to the recipients of the offer as the requirements of one or more of the provisions referred to in (i) or (ii);
- Any transaction entered into by the Company with a Bank as principal, on arms length terms and in the ordinary course of its banking business, as a result of which transaction such company has recourse to the credit risk of that Bank or in respect of a takeover offer by an Issuer that is a Code Company (as defined in the Takeovers Code).

Page 21: [43] Deleted	AL	13/06/2019 14:23:00
Nation of Masting relating to Dispagal or Assurbition of Assorb		

# Notice of Meeting relating to Disposal or Acquisition of Assets

The notice of meeting containing the resolution to approve any transaction referred to in clause 19.1 shall contain or be accompanied by such information, reports, valuations, and other material as are necessary to enable the holders of Securities to appraise the implications of the transactions.

Page 22: [44] Deleted	AL	13/06/2019 14:21:00
The text of any resolution to be put to a meeting for the purposes of clause 20.1 shall be set out		
in the notice of the releva	ant meeting. That notice shall:	

a. be approved by the Exchange in accordance with the Listing Rule 6.1;

b. be accompanied by an Appraisal Report; and

c. contain such other material as is necessary to enable the holders of Securities to decide whether the transaction price and terms are fair.

Page 24: [45] Del	eted	AL	13/06/2019 11:53:00
C.		0	t liable to retire by rotation by virtue of t in calculating the number of Directors

# Page 24: [46] Deleted AL 13/06/2019 11:53:00

# **Re-election of retiring Director**

A Director retiring by rotation at a meeting shall, if standing for re-election, be deemed to have been re-elected unless:

- a. some other person is elected to fill the vacated office; or
- b. it is resolved not to fill the vacated office; or
- c. a resolution for the re-election of that Director is put to the meeting and lost.

Page 26: [47] Deleted	AL	13/06/2019 12:02:00

# **EXECUTIVE DIRECTOR**

## Appointment

The Board may from time to time appoint a Director to the office of Executive Director (by whatever name called) for such period not exceeding five years, and on such terms, as the Board thinks fit. The Executive Director may be reappointed as Executive Director at the end of his or her term of such appointment.

Page 27: [48] Deleted	AL	13/06/2019 12:09:00

## Powers

The Board may from time to time entrust to and confer upon the Executive Director any of the powers exercisable by the Board upon such terms and conditions, and with such restrictions, as it may think fit and either collaterally with or to the exclusion of its own powers and may from time to time revoke, withdraw, alter or vary all or any of those powers and shall not affect the terms of the engagement of that Executive Director as an Employee.

# Not Subject to Retirement by Rotation

The Executive Director is not liable to retire by rotation under clause 22.6 while holding that office.

# Resignation

The Executive Director is subject to the same provision as regards resignation, removal and disqualification as the other Directors, and if the Executive Director ceases to hold the office of Director from any cause he or she automatically ceases to be Executive Director.

# Dismissal

Every Executive Director shall be liable to be dismissed or removed by the Board. The Board may enter into any agreement on behalf of the Company with any person who is, or is about to become, Executive Director, with regard to the terms and conditions of such person's employment, but so that the remedy of any such person for any breach of the agreement shall be in damages only, and such person shall have no right to claim to continue in such office contrary to the will of the Board.

# Page 27: [50] Deleted AL

# Remuneration

The remuneration of the Executive Director shall be fixed by the Board and may be in addition to the remuneration of that Executive Director as an ordinary Director.

13/06/2019 12:35:00

13/06/2019 12:09:00

Authorisation

Subject to the Listing Rules the Board may authorise the:

- a. payment of remuneration or the provision of other benefits by the Company to a Director for services in any capacity other than as a Director of the Company or a Subsidiary; or
- b. payment, notwithstanding the provisions of clause 25.3,

Page 27: [52] Deleted	AL	13/06/2019 12:35:00

- c. payment, notwithstanding the provisions of clause 25.3, of special remuneration to any non-executive Director who is or has been engaged by the Company to carry out any services which in the opinion of the Board are additional to those usually required of non-executive directors of similar companies in accordance with the requirements of the Listing Rules; or
- d. payment by the Company to a Director or former Director of compensation for loss of office other than as a Director; or
- e. making of loans by the Company to a Director; or
- f. giving of guarantees by the Company for debts incurred by a Director; or
- g. entering into of a contract to do any of the things set out in subclauses a. to f. of this clause 25.1;

if the Board is satisfied that to do so is fair to the Company.

Page 27: [53] Deleted	AL	13/06/2019 12:37:00
Interests Register and Certificate		

If a payment, benefit, loan, guarantee or contract is authorised under clause 25.1:

- a. the Board must ensure that particulars thereof are immediately entered in the interests register; and
- b. Directors who vote in favour thereof must sign a certificate stating that, in their opinion, it is fair to the Company, and the grounds for that opinion.

Page 27: [54] Deleted	AL	13/06/2019 12:38:00

# Shareholder Approval

No remuneration shall be paid to a Director in his or her capacity as Director of the Company or any Subsidiary other than a Subsidiary which is Listed unless that remuneration has been authorised by an Ordinary Resolution. Each such resolution shall express the Director's remuneration as either:

- a. a monetary sum per annum payable to all Directors of the Company taken together; or
- b. a monetary sum per annum payable to any person from time to time holding office as a Director of the Company.

Page 27: [55] Deleted         AL         13/06/2019 12:40:00
--------------------------------------------------------------

# Power to Increase when Additional Director

lf:

- a. the approved remuneration of the Directors is expressed in accordance with clause 25.3a; and
- b. the total number of Directors of the Company holding office is increased;

then the amount of remuneration payable in accordance with clause 25.3.a may be increased by the Board without Shareholder approval by such amount as is necessary to enable the Company to pay the additional Director or Directors of the Company remuneration not exceeding the average amount then being paid to each of the other non-executive Directors (other than the chairperson).

Page 28: [56] Deleted	AL	13/06/2019 12:40:00
Netter of second the		

# Notice of proposed increase

No resolution which increases the amount of the Director's remuneration shall be approved at a meeting of the Company unless notice of the amount of increase has been given in the notice of meeting. Nothing in this clause shall affect the remuneration of Executive Directors in their capacity as executives.

# **Retirement benefits**

The Company may make a payment to a Director or former Director of the Company, or to his or her dependants, by way of a lump sum or pension, upon or in connection with the retirement or cessation of office of that Director, only if the amount of the payment or the method of calculation of that payment is authorised by an Ordinary Resolution of the Company provided that the Company may make payment to a Director or former Director if such payment is permitted by the Listing Rules but nothing in this clause shall affect any amount paid to an Executive Director upon or in connection with the termination of his or her employment with the Company, or the payment of any amount attributable to the contribution (or any normal subsidy related thereto) made by a Director to a superannuation scheme.

Page 29: [57] Delete	d	AL	13/06/2019 12:44:00
In this cla	ause 26:		
Director.	"Director"	includes a former Director ar	nd " <b>director</b> " includes a former
insuranc		" includes pay, whether direct	ly or indirectly, the costs of the
	"Employee"	includes a former employee.	
"Indemnify" includes relief or excuse from liability, whether before or after the liability arises; and "Indemnity" and "Indemnified" have corresponding meanings.			
Page 31: [58] Delete	d	AL	11/06/2019 20:40:00
Preference on Liquidation to Series A Convertible Preference Shares			
	a tha avant of tha	liquidation or winding up of	the Company the holders of

In the event of the liquidation or winding up of the Company, the holders of series A convertible preference Shares will be entitled to receive cash in

preference to any other class of Shareholder up to an amount equal to the issue price of the series A convertible preference Shares.

An amalgamation of the Company under sections 219 to 225 of the Act or a sale of all or substantially all of the Company's assets will be deemed to be a liquidation or winding up for the purposes of this clause 30.2.

Page 31: [59] Deleted	AL	13/06/2019 13:29:00

# 

- The Company shall, as soon as reasonably practicable after the adoption of this 31.1 Constitution but not later than the Listing Rules so prescribe, have an Audit Committee which shall:
  - a. be comprised of solely Directors of the Company; and
  - have at least three members; and b.
  - have a majority of the members that are Independent Directors; and C.
  - d. have at least one member with an accounting or financial background.
- 31.2 The responsibilities of the Audit Committee include as a minimum:
  - ensuring that the processes are in place and monitoring those processes so a. that the Board is properly and regularly informed and updated on corporate financial matters; and
  - b. recommending the appointment and removal of the independent auditor; and
  - meeting regularly to monitor and review the independent and internal auditing c. practises; and
  - having direct communication with and unrestricted access to the independent d. and any internal auditors or accountants; and
  - reviewing the financial reports and advising all Directors whether they comply e. with the appropriate laws and regulations; and
  - f. ensuring that the external auditor or lead audit partner is changed at least every five years.

Page 34: [60] Deleted	AL	13/06/2019 13:37:00
No quorum		

## No quorum

If a quorum is not present within 30 minutes after the time appointed for the meeting:

- a. in the case of a meeting called under section 121(b) of the Act, the meeting is dissolved:
- b. in the case of any other meeting, the meeting is adjourned to the same day in the following week at the same time and place, or to such other date, time, and place as the Directors may appoint, and, subject to the Constitution of the Company, if, at the adjourned meeting, a quorum is not present within 30

minutes after the time appointed for the meeting, the Shareholders or their proxies present are a quorum.